## 01-06-2004

DEPARTMENT OF COMMERCE RECOF Form PTO-1594 J.S. Patent and Trademark Office (Rev. 10/02) OMB No. 0651-0027 (exp. 6/30/2005) 102638996 To the Honorable Commissioner of Patents and Trademarks: Please record the attached original documents or copy thereof. 2. Name and address of receiving party(ies) 1. Name of conveying party(ies): Nationwide Furniture Rentals Name: Mathress Giant Holding Corp.)
Internal a Delaware corporation 4 Sales, Inc., a Pennsylvania cap. Address: Association Individual(s) Street Address: General Partnership Limited Partnership State: Corporation-State Other Individual(s) citizenship\_\_\_\_ Association\_ Additional name(s) of conveying party(ies) attached? Types I No General Partnership\_ 3. Nature of conveyance: Limited Partnership Assignment Merger Corporation-State Delaware Change of Name Security Agreement Other\_ If assignee is not domiciled in the United States, a domestic Other representative designation is attached: 📮 Yes 🗣 No Execution Date:\_ 4. Application number(s) or registration number(s): B. Trademark Registration No.(s) A. Trademark Application No.(s) 2,176,086 Additional number(s) attached Yes No 5. Name and address of party to whom correspondence 6. Total number of applications and concerning document should be mailed: registrations involved: ..... Name: Lee Hone Le Blonc, Es 7. Total fee (37 CFR 3.41).....\$ Internal Address: Enclosed Authorized to be charged to deposit account 8. Deposit account number: Street Address: 1835 E. Hallandale City: Hallandale State: F1. Zin: 33009 (Attach duplicate copy of this page if paying by deposit account) DO NOT USE THIS SPACE 9. Statement and signature. To the best of my knowledge and belief, the foregoing information is true and correct and any attached copy is a true

Mail documents to be recorded with required cover sheet information to:

Commissioner of Patent & Trademarks, Box Assignments

Washington, D.C. 20231

01/05/2004 DBYRNE

00000167 2176086

copy of the original document.

Lee Anne LeBlanc

Name of Person Signing

01 FC:8521

40.00 OP

TRADEMARK REEL: 002890 FRAME: 0877

## AGREEMENT AND PLAN OF MERGER

by and among

THE MG MERGER ENTITIES,

MATTRESS GIANT HOLDING CORPORATION,

MATTRESS GIANT ACQUISITION CORPORATION,

THE NATIONWIDE ENTITIES and

THE SHAREHOLDERS OF THE NATIONWIDE ENTITIES

Dated as of February 19, 1999

TRADEMARK
REEL: 002890 FRAME: 0878

# AGREEMENT OF PLAN AND MERGER

THIS AGREEMENT AND PLAN OF MERGER (the "Agreement") is entered into as of February 19, 1999 by and among Mattress Giant Holding Corporation, a Delaware corporation ("MGHC"), Mattress Giant Acquisition Corporation, a Delaware corporation (the "Parent"), MG Merger Sub No. 1, Inc., a Pennsylvania corporation ("Sub No. 1"), MG Merger Sub No. 2, Inc., a Pennsylvania corporation ("Sub No. 2"), Nationwide Furniture Rentals & Sales, Inc., a Pennsylvania corporation ("NFR"), Nationwide Mattress Giant, Inc., a Pennsylvania corporation ("Nationwide") and each of the persons listed on Exhibit A attached hereto who constitute all of the holders of all of the issued and outstanding capital stock of the Nationwide Entities, (such person are referenced to herein individually as a "Shareholder" and, collectively as the "Shareholders"). Sub No. 1 and Sub No. 2 are sometimes referred to herein individually as a "MG Merger Entity" and collectively as the "MG Merger Entities". NFR and Nationwide are sometimes referred to herein individually as a "Nationwide Entity" and collectively as the "Nationwide Entities."

# **WITNESSETH**

- A. MGHC owns all of the issued and outstanding shares of capital stock of the Parent. The Parent owns all of the outstanding shares of capital stock of the MG Merger Entities.
- B. The Board of Directors of each of the MG Merger Entities has determined that it is fair to, and in the best interests of, their respective corporations and stockholders for the MG Merger Entities to be merged with and into the Nationwide Entities upon the terms and subject to the conditions set forth herein.
- C. The Board of Directors of each of the MG Merger Entities, the Parent and the Nationwide Entities have approved the Merger Transactions (as defined in Section 1.1 herein) in accordance with the General Corporation Law of the State of Delaware and the Business Corporation Law of 1988 of the Commonwealth of Pennsylvania (the "Pennsylvania Law"), as applicable.
- D. The Nationwide Entities are engaged in the business of operating a chain of 47 retail mattress stores, along with related warehousing, inventory and delivery services, in the States of Pennsylvania, Delaware and New Jersey (the "Business").

NOW THEREFORE, in consideration of the mutual covenants, agreements, representations and warranties set forth herein, the parties hereto agree as follows:

# ARTICLE 1. THE MERGER TRANSACTION.

1.1 <u>Merger</u>. Upon the terms and subject to the conditions of this Agreement, at the Effective Time Sub No. 1 shall be merged with and into NFR ("Merger No. 1") and Sub No.2 shall be merged with and into Nationwide ("Merger No. 2"), each in accordance with the applicable provisions of the Pennsylvania Law. Merger No. 1 and Merger No. 2 are referred to herein

1

STARKP/AGR/134740\_7(16480/03)

TRADEMARK
REEL: 002890 FRAME: 0879

individually as a "Merger Transaction" and collectively as the "Merger Transactions." Each of the MG Merger Entities and the Nationwide Entities are herein sometimes referred to as the "Constituent Corporations." At the Effective Time, the identity and separate corporate existence of each respective MG Merger Entity shall cease and each of the respective Nationwide Entities shall be the surviving corporation in the Merger Transactions (sometimes hereinafter referred to as the "Surviving Corporations").

1.2 <u>Effective Time</u>. The Merger Transactions shall become effective on the date and at the time the Articles of Merger referred to in Section 1.9 hereof are filed with the Secretary of State of the Commonwealth of Pennsylvania in accordance with Section 1927 of the Pennsylvania Law, which filing shall be made as soon as practicable after the Closing. The time at which the Merger Transactions shall become effective as aforesaid is referred to hereinafter as the "Effective Time." The Merger Transaction shall have the effect set forth in Section 1929 of the Pennsylvania Law.

# 1.3 Articles of Incorporation, Bylaws, Directors and Officers, Name.

- (a) The Articles of Incorporation of each of Sub. No. 1 and Sub No. 2, as in effect immediately prior to the Effective Time, shall be the Articles of Incorporation of the respective Surviving Corporation.
- (b) The Bylaws of each of Sub No. 1 and Sub No. 2, as in effect immediately prior to the Effective Time, shall be the Bylaws of the respective Surviving Corporation.
- (c) The directors and officers of each of Sub No. 1 and Sub No. 2, in office immediately prior to the Effective Time, shall be the directors and officers of the respective Surviving Corporation, and each shall hold his or her respective office or offices from and after the Effective Time until his or her successor shall have been elected and shall have been qualified.
- (d) The name of each respective Surviving Corporation from and after the Effective Time shall be the name of the respective Nationwide Entity, (i.e. Nationwide Furniture Rental and Sales, Inc. and Nationwide Mattress Giant, Inc.) and the Articles of Incorporation of each of Sub No. 1 and Sub No. 2 shall be amended to reflect such name.
- 1.4 Assets and Liabilities. At the Effective Time, the Surviving Corporations shall possess all the rights, privileges, powers and franchises of a public as well as of a private nature, and be subject to all the restrictions, disabilities and duties of each of the Constituent Corporations; and all the rights, privileges, powers and franchises of each of the Constituent Corporations, and all property, real, personal and mixed, and all debts due to any of the Constituent Corporations on whatever account shall be vested in each respective Surviving Corporation of each Merger Transaction; and all property, rights, privileges, powers and franchises, and all and every other interest shall be thereafter as effectually the property of each respective Surviving Corporation as they were of the several and respective Constituent Corporations; but all rights of creditors and all

2

liens upon any property of any of the Constituent Corporations shall be preserved unimpaired, and all debts, liabilities and duties of the respective Constituent Corporations shall thenceforth attach to the appropriate Surviving Corporation of each Merger Transaction, and may be enforced against it to the same extent as if said debts, liabilities and duties had been incurred or contracted by it.

Corporations shall consider or be advised that any deeds, bills of sale, assignments or assurances or any other acts or things are necessary, desirable or proper (i) to vest, perfect or confirm, of record or otherwise, in the respective Surviving Corporation of each Merger Transaction, its right, title or interest in, to or under any of the rights, properties or assets of the respective Constituent Corporation acquired or to be acquired as a result of the Merger Transactions, or (ii) otherwise to carry out the purposes of this Agreement, the affected Surviving Corporation and its proper officers and directors or their designees shall be authorized to execute and deliver, in the name and on behalf of the Constituent Corporations, all such deeds, bills of sale, assignments and assurances and do, in the name and on behalf of the Constituent Corporations, all such other acts and things necessary, desirable or proper to vest, perfect or confirm its right, title or interest in, to or under any of the rights, properties or assets of the Constituent Corporations acquired or to be acquired as a result of the Merger Transactions and otherwise to carry out the purposes of this Agreement.

## 1.6 Conversion of Securities.

- (a) By virtue of the Merger Transactions and without any action on the part of the holder thereof, at the Effective Time, all of the shares of capital stock of the Nationwide Entities in the aggregate issued and outstanding immediately prior to the Effective Time (the "Nationwide Stock") shall be converted into the right to receive, and become exchangeable for \$2,450 per share, or an aggregate of \$24,500,000 (the "Merger Consideration") to be paid in cash to each Shareholder based on and in proportion to their holdings of the capital stock of the Nationwide Entities as set forth on Exhibit B.
- the right in the aggregate to receive, in lieu of up to \$4,500,000 in cash, shares of common stock of MGHC ("MGHC Stock") at a price of \$50 per share, subject to the following conditions: (i) any Shareholder electing to receive a portion of the Merger Consideration payable to him in MGHC Stock must give MGHC and the Parent written notice of his decision ("Election Notice") specifying the dollar amount of the Merger Consideration to be paid by shares of MGHC Stock, together with a completed and executed Investment Representation Letter and a Consent to Shareholders Agreement in the form of Exhibits C and D, respectively, at least ten (10) days prior to the Closing; and (ii) if the Shareholders elect to receive in the aggregate more than \$4,500,000 of the Merger Consideration in MGHC Stock, then each Shareholder giving an Election Notice shall be entitled to and shall receive that number of MGHC Stock specified in his Election Notice multiplied by a percentage equal to the percentage provided by the product of \$4,500,000 divided by the

3

aggregate amount of Merger Consideration specified to be converted in all of the Election Notices properly delivered hereunder.

- (c) By virtue of the Merger Transactions and without any action on the part of the holder thereof, at the Effective Time, each share of capital stock held by any of the Nationwide Entities as a treasury share immediately prior to the Effective Time shall be canceled and no payment of any consideration shall be made with respect thereto.
- (d) By virtue of the Merger Transactions and without any action on the part of the holder thereof, at the Effective Time, each share of common stock of each of the MG Merger Entities that is issued and outstanding immediately prior to the Effective Time shall be converted into one newly issued share of Common Stock of the respective Surviving Corporation.
- 1.7 <u>Exchange of Shares</u>. At the Closing each Shareholder shall deliver to the Parent all certificates representing all of the Nationwide Stock owned by him.

# 1.8 Payment of Merger Consideration.

- (a) At the Closing, but in no event later than the Effective Time, MGHC and Parent shall jointly and severally pay and make available to, and each Shareholder will be entitled to receive, upon surrender to the Parent of one or more certificates representing his shares of Nationwide Stock for cancellation in accordance with the procedures set forth in Section 1.8(b) hereof, the Merger Consideration into which the outstanding Nationwide Stock are converted in the Merger. The MGHC Stock into which the shares of Nationwide Stock are converted, if any, shall be deemed to be issued at the Effective Time.
- (b) At the Closing, or at such later time that a Shareholder shall surrender certificate(s) evidencing his Nationwide Stock to Parent pursuant to the same method as provided by the giving of notices hereunder, Parent and MGHC shall jointly and severally pay and deliver the Merger Consideration to the Shareholders who have so surrendered certificates of Nationwide Stock by (i) with respect to cash consideration, a wire transfer to an account set forth on Exhibit B as to each such Shareholder or as otherwise specified in subsequent written notice from Shareholder to Parent; and (ii) with respect to MGHC Stock, a certificate representing that number of shares of MGHC Stock to which the Shareholder is entitled to receive pursuant to the terms hereof and issued in the name of such Shareholder. Until surrendered as contemplated by this Section, each certificate shall be deemed at any time after the Effective Time to represent only the right to receive, upon such surrender, the Merger Consideration.
- (c) As of the date hereof, the stock transfer books of the Nationwide Entities shall be closed and no transfer of shares of Nationwide Stock shall thereafter be made. If after the Closing, certificates representing such shares of Nationwide Stock are presented to the

4

Surviving Corporation, they shall be cancelled and exchanged for the Merger Consideration in accordance with the terms of this Agreement.

- (d) All shares of MGHC Stock to be issued in accordance with the terms of this Agreement, if any, shall be "restricted securities" as that term is defined pursuant to the Securities Act of 1933, as amended, and the rules and regulations promulgated thereon ("1933 Act"). Certificates representing MGHC Stock, when issued, shall contain a restrictive legend to the effect that the MGHC Stock have not been registered pursuant to the 1933 Act and may not be sold, transferred or otherwise disposed of except in compliance with the 1933 Act or unless MGHC receives an opinion of counsel that an exemption from registration is available.
- (e) If any certificate representing Nationwide Stock is lost, stolen or destroyed and cannot be delivered at the Closing by the Shareholder, upon the making of an affidavit of the fact by the Shareholder claiming such certificate to be lost, stolen and destroyed and, if required by the Surviving Corporation, the posting by such person of a bond in such reasonable amount as the Surviving Corporation may direct as indemnity against any claim that may be made against it with respect to such certificate, Parent and MGHC shall issue in exchange for such lost, stolen or destroyed certificate, the Merger Consideration pursuant to this Agreement.
- (f) Notwithstanding any provision in this Section to the contrary, a portion of the Merger Consideration payable in cash to each Shareholder hereunder shall be paid into escrow pursuant to Section 1.12(b) and 9.2(a) hereto in an amount equal to 1.02% and 8%, respectively, of the Merger Consideration payable to such Shareholder.
- 1.9 <u>Requisite Shareholder Approval</u>. Each of the Nationwide Entities will take all action necessary in accordance with applicable law and their Articles of Incorporation and Bylaws to convene a special meeting of the Shareholders to consider and vote upon, or solicit the written consent of the Shareholders for, the approval of the Merger Transactions and adoption of this Agreement. Each of the Nationwide Entities hereby represents and warrants to the Parent that the approval of the Merger Transactions and the adoption of this Agreement will require, under the Pennsylvania Law and the Articles of Incorporation of each of the Nationwide Entities, the approval of the holders of a majority of the outstanding shares of common stock.
- 1.10 <u>Consummation of Merger</u>. As soon as practicable after satisfaction of the conditions set forth in Article 6 hereof, each of the Nationwide Entities and the MG Merger Entities shall execute and deliver to the Secretary of State of Pennsylvania, duly executed and verified Articles of Merger in the form of Exhibit E attached hereto (the "Articles of Merger").
- 1.11 <u>Time and Place of Closing</u>. The closing of the Merger Transactions and other transactions contemplated or provided for in this Agreement (herein called the "Closing") shall be held at the offices of Kane, Russell, Coleman & Logan, P.C., 3700 Thanksgiving Tower, 1601 Elm

5

Street, Dallas, Texas 75201, on March 22, 1999 (the "Closing Date") or at such other place or earlier or later date or time as may be fixed by mutual agreement of the Parent and the Shareholders.

### 1.12 Closing Equity.

- (a) The Merger Consideration has been calculated and agreed to based on the representation and warranty that the total stockholders' equity of all Nationwide Entities as of December 31, 1998 as set forth in the 1998 Balance Sheet (as defined below) will equal at least \$186,000 (the "1998 Agreed Equity"). As the Closing Date shall occur after December 31, 1998, the Shareholders agree, jointly and severally, that they shall not make or declare any dividends or distributions of any nature based on and in respect of the equity of the Nationwide Entities, except as expressly provided in this Agreement.
- In the event the actual total stockholder's equity of the Nationwide Entities (b) as of December 31, 1998 (the "1998 Actual Equity") exceeds the 1998 Agreed Equity by an amount in excess of \$50,000, the Parent will pay to the Shareholders, in proportion to their holding of the common stock of the Nationwide Entities as set forth on Exhibit "B", an aggregate amount equal to the difference between (i) the 1998 Actual Equity and (ii) the 1998 Agreed Equity plus \$50,000. In the event the 1998 Actual Equity is less than the 1998 Agreed Equity by an amount in excess of \$50,000, the Shareholders, in proportion to their holding of the common stock of the Nationwide Entities as set forth on Exhibit "B", agree to pay to the Parent an amount equal to the difference between (i) the 1998 Actual Equity plus \$50,000 and (ii) the 1998 Agreed Equity. Furthermore, the Shareholders agree to escrow at the Closing, in proportion to their holding of the common stock of the Nationwide Entities as set forth on Exhibit B, an aggregate of \$250,000 of the Merger Consideration (the "Escrow Deposit") pursuant to an Escrow Agreement substantially in the form attached hereto as Exhibit F, such escrow to guarantee the performance of the Shareholders as contemplated in the prior sentence. The term "total stockholder's equity" as utilized in subsection 1.12(a) and this subsection 1.12(b) shall mean total assets less total liabilities of the Nationwide Entities and shall be calculated in accordance with generally accepted accounting principles applied consistent with the total stockholders' equity set forth in the audited combined balance sheets of December 31, 1997 as audited by Stockton & Bates in their report dated October 22, 1998. The obligations of the Shareholders pursuant to this Section shall be defined as a Parent Indemnifiable Loss under Section 9.1 hereof but shall not apply or be counted against the Minimum Threshold under Section 9.2 hereof.
- (c) The Nationwide Entities, acting through the Shareholders' Representative (as defined in Section 1.13 hereof), will cause Stockton & Bates (the "Nationwide Accountants") to commence reviewing the books and records of each of the Nationwide Entities as of December 31, 1998 and in connection therewith prepare an audited combined balance sheet of the Nationwide Entities as of such date based on generally accepted accounting principles, consistently applied (herein, collectively, the "1998 Balance Sheet") and related audited statements of income, stockholders equity and cash flow. If the Nationwide Accountants

6

have not completed and delivered the audited 1998 Balance Sheet prior to the Closing, the Parent and MGHC shall request the Nationwide Accountants to promptly complete the audit of the 1998 Balance Sheet as soon as reasonably practicable after the Closing. The party to whom the Nationwide Accountants shall deliver the 1998 Balance Sheet (the "Sender") will promptly deliver to all other parties (the "Recipient") a copy of the 1998 Balance Sheet and such audit shall be deemed accepted by the Recipient (in the case the Recipients are the Shareholders, then they shall act through the Representative) unless the Sender (in the case the Sender is the Shareholders, then they shall act through the Representative) shall have received, within thirty (30) days after delivery thereof, a written notice from the Recipient of the Recipient's objections to the 1998 Balance Sheet. In the event the Recipient timely objects to the 1998 Balance Sheet, the Recipient shall have the right to retain independent accountants of their selection and at their expense, during the ninety (90) day period following expiration of the thirty (30) day period above, to review the books and records of the Nationwide Entities and the work papers of the Nationwide Accountants to ascertain whether there are any discrepancies in the preparation of the 1998 Balance Sheet. The Recipient shall supply to the Sender a copy of any report generated by the Recipient's independent accountants. Until the audit and subsequent reviews contemplated by this subsection 1.12(c) are completed, the Shareholders' Representative, his agents and accountants and the Nationwide Accountants shall have full and complete access to the books and records of the Nationwide Entities. The Parent agrees to keep the books and records of the Nationwide Entities in Philadelphia, Pennsylvania through April 15, 1999.

- (d) In the event there shall continue to remain any dispute regarding the 1998 Actual Equity following the examination by the Recipient's independent accountants, such dispute shall be submitted for arbitration by the American Arbitration Association (the "AAA") in Philadelphia, Pennsylvania, the expense of which shall be shared equally by the Parent and the Shareholders, provided that each party shall bear its own expenses. The parties shall request the AAA to appoint three arbitrators for such dispute whose decision shall be final.
- (e) Any amounts due pursuant to subsection 1.12(b) shall be paid within ten days of (i) the Recipient's acceptance or deemed acceptance of the 1998 Balance Sheet, (ii) agreement among the parties as to the amount due under subsection 1.12(b), or (iii) the arbitrators' decision as contemplated in subsection 1.12(d).
- (f) Notwithstanding anything to the contrary contained herein, in the event (i) the 1998 Balance Sheet reflects total stockholders' equity of the Nationwide Entities of less than \$136,000 and (ii) such 1998 Balance Sheet is delivered to the Nationwide Entities prior to the Closing Date, the Shareholders may terminate this Agreement pursuant to Article 7 hereof.

## 1.13 Shareholders' Representative.

- (a) By the execution and delivery of this Agreement, each Shareholder, for himself, his heirs, executors, legal representatives and successors and assigns, hereby irrevocably constitutes and appoints Kenneth F. Mazda as his or her true and lawful agent and attorney-in-fact referred to as (the "Representative"), with full power of substitution to act in his, her or its name, place and stead with respect to all transactions contemplated by and all terms and provisions of this Agreement, and to act on his, her or its behalf in any dispute, litigation or arbitration involving matters arising as a result of and after the Closing under this Agreement, the Escrow Agreement referred to in Section 1.12(b), the Deposit Agreement referred to in Section 10.10, and the Indemnity Escrow Agreement referenced to in Section 9.2(a) and to take all actions on behalf of the Shareholders in connection therewith.
- (b) The appointment of the Representative shall be deemed coupled with an interest and shall be irrevocable, and the Parent and any other person may conclusively and absolutely rely, without inquiry, upon any action of each Representative on behalf of the Shareholders in all matters in which he has been granted authority pursuant to Section 1.13(a). The Representative shall act for the Shareholders on all of the matters set forth in this Agreement in the manner the Representative believes to be in the best interest of the Shareholders whom he represents and consistent with his obligations under this Agreement, but the Representative shall not be responsible to the Shareholders for any loss or damages the Shareholders may suffer by reason of the performance by the Representative of his duties under this Agreement, other than loss or damage arising from fraud in the performance of his duties under this Agreement.
- (c) All actions, decisions and instructions of the Representative taken, made or given pursuant to the authority granted to such Representative pursuant to subsection 1.12(a) above shall be final, conclusive and binding upon all Shareholders and all actions, decisions and instructions of the Representative taken, made or given pursuant to the authority granted to such Representative pursuant to subsection 1.13(a) above shall be conclusive and binding upon all individual Shareholders. The Parent, the MG Merger Entities, their officers, directors, employees, agents and affiliates shall be able to rely exclusively on the instructions, decisions and actions of the Representative.
- (d) The provisions of this Section 1.13 are independent and severable, shall constitute an irrevocable power of attorney, coupled with an interest and surviving death or dissolutions, granted by the Shareholders to the Representative and shall be binding upon the executors, heirs, legal representatives, successors and assigns of each such Shareholder.
- (e) Each of the Shareholders, jointly and severally, hereby agrees to indemnify and hold harmless the Representative of and from any and all losses, liabilities, obligations, damages, deficiencies, actions, suits, proceedings, demands, assessment, orders judgments,

8

fines, penalties, costs and expenses of any kind or nature whatsoever (each, a "Representative Loss") sustained, suffered or incurred by Representative arising out of his actions as the Representative unless such Representative Loss is caused by the fraud of the Representative.

- (f) The Representative, in his capacity as such, shall have no liability to the Parent or MGHC for any failure to perform any of his obligations hereunder; and any failure of the Representative to respond or answer shall be deemed a rejection or negative response of the Representative. If Kenneth F. Mazda dies or otherwise becomes incapacitated and is unable to serve as the Representative, then and in such event all references herein to the Representative shall be deemed to be the Shareholders.
- 1.14 <u>Allocation</u>. The Shareholders, the Nationwide Entities, the Parent and the MG Merger Entities agree to establish a purchase price allocation as set forth on Exhibit G attached hereto. The parties agree to file all tax returns and information reports for tax reporting purposes in a manner consistent with such allocation.
- 1.15 <u>HSR Filing</u>. The Parent and MGHC have agreed with the Shareholders that no filing is required under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, based on the representation and warranty of the Nationwide Entities and the Shareholders that the total assets of the Nationwide Entities as of December 31, 1998 was less than Ten Million Dollars (\$10,000,000).

## 1.16 Conduct of Closing.

- (a) At or prior to the Closing, the Nationwide Entities and the Shareholders shall deliver to Parent:
- (i) Articles of Merger in the form of Exhibit E hereto ("Articles of Merger") duly executed by the Nationwide Entities;
  - (ii) The certificates required by Section 6.1(a) and 6.1(k) hereof;
- (iii) The Escrow Agreement and Indemnity Escrow Agreement duly executed by the Shareholders;
  - (iv) The Non-Compete Agreements duly executed by the Shareholders;
- (v) Amendments to the Leases for the warehouse facilities in the forms set forth collectively in Exhibit H hereto;
  - (vi) The opinion of counsel referred to in Section 6.1(h);

9

- (vii) A resignation of each of the Shareholders as an officer and director of the Nationwide Entities;
- (viii) A Warranty and Maintenance Agreement in the form of Exhibit I duly executed by the Shareholders; and
- (ix) Any other documents and certificates contemplated by Section 6.1 hereof.
- (b) At or prior to the Closing, MGHC and the Parent shall deliver to the Shareholders:
  - (i) Articles of Merger duly executed by the MGHC Merger Entities;
- (ii) The Merger Consideration to the Shareholders who have tendered their certificates for their Nationwide Stock;
- (iii) The Escrow Agreement and Indemnity Escrow Agreement duly executed by Parent; and
- (iv) The certificates and opinion of counsel referred to in Section 6.2 hereof.
- (c) Notwithstanding the foregoing, the Nationwide Entities and the Shareholders shall not be liable for a breach of this Agreement due to the failure of any Shareholder to deliver any of the amendments and instruments set forth above in Section 1.16(a); provided, however, MGHC and Parent, in any such event, may terminate this Agreement.

# ARTICLE 2. REPRESENTATIONS AND WARRANTIES OF THE NATIONWIDE ENTITIES AND THE SHAREHOLDERS.

2.1 <u>Making of Representations and Warranties</u>. As a material inducement to MGHC, the Parent and the MG Merger Entities to enter into this Agreement and to consummate the transactions contemplated hereby each of the Nationwide Entities and each of the Shareholders, jointly and severally, hereby make to MGHC, the Parent and the MG Merger Entities the representations and warranties contained in this Article 2 as of the date hereof; provided that the representations and warranties in subsections 2.4 and 2.5(b) and (c) are only made by each Shareholder with respect to itself or himself and not with respect to any other Shareholder. All representations and warranties made in this Article 2 shall survive for a period of twelve (12) months after the Closing Date; save and except the representations and warranties made in Section 2.4 and Section 2.8 which shall survive for a period of forty-eight (48) months after the Closing Date.

10

- Organization and Qualifications of the Company. Each of the Nationwide Entities 2.2 is a corporation duly organized, validly existing and in good standing under the laws of the state set forth in Schedule 2.2 with full corporate power and authority to own or lease its properties and to conduct its business in the manner and in the places where such properties are owned or leased and where such business is currently conducted or proposed to be conducted. The copies of each Nationwide Entity's articles of incorporation, as amended to date, and of each Nationwide Entity's by-laws, as amended to date, and heretofore delivered to the Parent's counsel, are complete and correct, and no amendments thereto are pending. Each of the Nationwide Entities is duly qualified to do business as a foreign corporation in those states listed on Schedule 2.2, and neither is required to be licensed or qualified to conduct its business or own its property in any other jurisdiction in which the failure to be so qualified would have a material adverse effect on the such entities' properties, assets, prospects, financial condition or business or on the Business as a whole. The copies of the minute books and stock records of each Nationwide Entity heretofore delivered to the Parent's counsel are complete and correct copies of all such records maintained by the Nationwide Entities.
- 2.3 <u>Corporation Subsidiaries</u>. Except as set forth in Schedule 2.3, no Nationwide Entity owns any equity interests in any other entity.

## 2.4 *Equity*.

- (a) The total authorized capital stock of each Nationwide Entity is set forth in Schedule 2.4(a). Schedule 2.4(a) contains a true, correct and complete listing of the stockholders of each Nationwide Entity at the date hereof, together with the number of shares and class of capital stock owned therein by each such stockholder. Except as set forth in Schedule 2.4(a), there are no outstanding subscriptions, options, warrants, commitments, preemptive rights, agreements, arrangements or commitments of any kind for or relating to the issuance, sale, registration or voting of, or outstanding securities convertible into or exchangeable for, any shares of capital stock of any class or other equity interests of any Nationwide Entity.
- (b) Each Shareholder owns beneficially and of record the number of shares of stock in each Nationwide Entity set forth opposite such Shareholder's name in Schedule 2.4(a), free and clear of all pledges, liens, security interests, restrictions, encumbrances or other claim or charge except as specifically set forth in such schedule.

# 2.5 <u>Authority of the Nationwide Entities and the Shareholders.</u>

(a) Each Nationwide Entity has full corporate right, power and authority to enter into this Agreement and each agreement, document and instrument to be executed and delivered by it pursuant to the terms of this Agreement. The execution, delivery and performance by each Nationwide Entity of this Agreement and each such other agreement, document and instrument has been duly authorized by all necessary action of each Nationwide Entity and its stockholders. This Agreement and each agreement, document and instrument to be executed and delivered by each

11

Nationwide Entity pursuant to, or as contemplated by, this Agreement constitute, or will when executed and delivered constitute, valid and binding obligations of each Nationwide Entity, enforceable in accordance with their respective terms. The execution, delivery and performance by each Nationwide Entity of this Agreement and each such agreement, document and instrument pursuant to, or as contemplated by, this Agreement to which it is a party:

- (i) do not and will not violate any provision of the articles of incorporation or by-laws, as applicable, of each Nationwide Entity;
- (ii) except as set forth on <u>Schedule 2.5</u>, do not and will not violate any laws of the United States, or any state or other jurisdiction applicable to any of the Nationwide Entities or require any of the Nationwide Entities to obtain any approval, consent or waiver of, or make any filing with, any person or entity (governmental or otherwise) that has not been obtained or made, and except for any other approvals, consents, waivers and filings that, if not obtained or made, individually or in the aggregate would not have an adverse effect on the properties, assets, business or condition (financial or other) of the Nationwide Entities or the Business; and
- (iii) except as set forth on Schedule 2.5, do not and will not result in a breach of, constitute a default under, accelerate any obligation under, or give rise to a right of termination of any indenture or loan or credit agreement or any other agreement, contract, instrument, mortgage, deed of trust, lien, lease, permit, authorization, order, writ, judgment, injunction, decree, determination or arbitration award, whether written or oral, to which any Nationwide Entity is a party or by which the property (owned or subject to a lease) of any Nationwide Entity is bound or affected, except (A) where such breach, default, acceleration or right of termination would not have an adverse effect on the properties, assets, business, or condition (financial or other) of any Nationwide Entity and would not result in the creation or imposition of any mortgage, pledge, lien, security interest or other charge or encumbrance on any of the assets of any Nationwide Entity and (B) where a waiver or consent of such breach or default has been obtained.
- (b) If such Shareholder is an individual, such Shareholder has full right, authority, power and capacity to enter into this Agreement and each agreement, document and instrument to be executed and delivered by or on behalf of him pursuant to this Agreement. If such Shareholder is not an individual, such Shareholder has full right, power and authority to enter into this Agreement and each agreement, document and instrument to be executed and delivered by it pursuant to, or as contemplated by, this Agreement and to carry out the transactions contemplated hereby and thereby, including but not limited to the Merger Transactions, and the execution, delivery and performance by such Shareholder of this Agreement and each such other agreement, document and instrument has been duly authorized by all necessary corporate, trust or partnership action, as appropriate.

12

- (c) This Agreement and each agreement, document and instrument executed and delivered by each Shareholder pursuant to, or as contemplated by, this Agreement constitute, or when executed and delivered will constitute, the valid and binding obligations of such Shareholder enforceable in accordance with their respective terms. The execution, delivery and performance by such Shareholder of this Agreement and each such agreement, document and instrument:
  - (i) if such Shareholder is not an individual, do not and will not violate any provision of any relevant formation document of such Shareholder;
  - (ii) do not and will not violate any laws of the United States, or any state or other jurisdiction applicable to such Shareholder or require such Shareholder to obtain any approval, consent or waiver of, or make any filing with, any person or entity (governmental or otherwise) that has not been obtained or made; and
  - (iii) do not and will not result in a breach of, constitute a default under, accelerate any obligation under or give rise to a right of termination of any indenture or loan or credit agreement or any other agreement, contract, instrument, mortgage, lien, lease, permit, authorization, order, writ, judgment, injunction, decree, determination or arbitration award to which such Shareholder is a party or by which the property of such Shareholder is bound or affected, or result in the creation or imposition of any mortgage, pledge, lien, security interest or other charge or encumbrance on any of the Shareholders' or Nationwide Entities' assets or properties.
- (d) This Agreement and the consummation of all transactions contemplated hereby have been, or will be prior to Closing, approved by the Board of Directors and shareholders of each Nationwide Entity at a duly called and properly held meeting or by unanimous written consent in lieu of such meeting. No other corporate action or approval of any Nationwide Entity or the holders of any class of the equity thereof or securities convertible into or exercisable for such capital stock is required to consummate the transactions contemplated hereby other than as set forth herein.

# 2.6 <u>Real and Personal Property</u>.

- (a) <u>Real Property</u>. All of the real property leased by each Nationwide Entity is identified on <u>Schedule 2.6(a)</u> (herein referred to as the "Leased Real Property."). Neither of the Nationwide Entities, as of the Closing Date, will own any real property.
  - (i) <u>Status of Leases</u>. All leases of Leased Real Property are identified on <u>Schedule 2.6(a)</u>, and copies thereof have been delivered to the Parent and its counsel. The Nationwide Entities have, as against the Nationwide Entities, enforceable leasehold interests in the Leased Real Property, and, except as noted in Schedule 2.6(a), have occupied the premises comprising the Leased Real Property

13

since the date of inception in each lease. There are no mortgages, liens or encumbrances against such leasehold interests in the Leased Real Property. Neither the Shareholders nor any of the Nationwide Entities has received any notice of any material uncured default under any terms of said leases, nor has any event occurred which, with notice or the passage of time, or both, would give rise to such a default. To the knowledge of the Nationwide Entities and the Shareholders, the other party to each of said leases is not in default under any terms thereof and there is no event which, with notice or the passage of time, or both, would give rise to such a default.

- (ii) <u>Consents.</u> Except as set forth in <u>Schedule 2.6(a)</u>, no consent is required with respect to the transactions contemplated by this Agreement from the other parties to any lease of Leased Real Property.
- (iii) <u>Condition of Leased Real Property</u>. To the actual knowledge of any of the Shareholders, except as set forth in <u>Schedule 2.6(a)</u>, no material portion of the Leased Real Property (involving, in each instance, an amount in excess of \$10,000) is not performing the function for which it was designed.
- (iv) <u>Compliance with the Law</u>. Neither any of the Nationwide Entities nor any Shareholder has received any notice from any governmental authority of any violation of any law, ordinance, regulation, license or authorization issued with respect to any Leased Real Property that has not been heretofore corrected and to their knowledge, no such violation exists which could have an adverse affect on the operation or value of any Leased Real Property or the Business.
- (b) <u>Personal Property</u>. A complete description of all depreciable machinery and equipment with a purchase price, or total remaining lease payments, of \$10,000 or more of each Nationwide Entity is contained in <u>Schedule 2.6(b)</u> hereto. Except as specifically disclosed in <u>Schedule 2.6(b)</u>, each Nationwide Entity has good title to all of its personal property shown on its books and records. None of such personal property or assets owned by the Nationwide Entities is subject to any mortgage, pledge, lien, conditional sale agreement, security title, encumbrance or other charge except as specifically disclosed in <u>Schedule 2.6(b)</u>.

#### 2.7 Financial Statements.

- (a) The Nationwide Entities and the Shareholders have delivered to the Parent the following financial statements:
  - (i) Audited balance sheets of each of the Nationwide Entities as of December 31, 1997 certified by Stockton & Bates, independent public accountants (the "Audited Balance Sheet");

14

- (ii) Unaudited combined statement of earnings of each of the Nationwide Entities for the year ended December 31, 1997 prepared by management of the Nationwide Entities (the "1997 Statement of Earnings"); and
- (iii) Unaudited balance sheet of each of the Nationwide Entities as of September 30, 1998 (the "Base Balance Sheet") and unaudited combined statement of earnings of the Nationwide Entities for the nine months then ended (collectively with the Base Balance Sheet, the "1998 Financial Statements").

Except as set forth on Schedule 2.7(a), said financial statements have been prepared in accordance with generally accepted accounting principles ("GAAP") applied consistently during the periods covered thereby (except that the 1998 Financial Statements are subject to normal year-end audit adjustments) and present fairly in all material respects the financial condition of the Nationwide Entities at the dates of said statements and the results of its operations for the periods covered thereby.

(b) As of the date hereof, the Nationwide Entities had no material liabilities of any nature, whether accrued, absolute, contingent or otherwise (including without limitation liabilities as guarantor or otherwise with respect to obligations of others, or liabilities for taxes due or then accrued or to become due or contingent or potential liabilities relating to activities of the Nationwide Entities or the conduct of its Business prior to the date hereof or the Closing, as the case may be), except liabilities (i) stated or adequately reserved against on the Base Balance Sheet or the notes thereto, (ii) reflected in the Schedules furnished to the Parent under this Agreement on the date hereof including Schedule 2.7(b), or (iii) incurred in the ordinary course of business of the Nationwide Entities consistent with the terms of this Agreement.

#### 2.8 Taxes.

- (a) A list of all federal, state and local income tax and information returns (other than personal property tax returns) filed with respect to the Nationwide Entities since December 31, 1996 is set forth in Schedule 2.8 attached hereto. For every taxable period of the Nationwide Entities, each Nationwide Entity has delivered to the Parent correct and complete copies of all federal and state income tax and information returns, examination reports and statements of deficiencies assessed against or agreed to by each Nationwide Entity. Schedule 2.8 attached hereto sets forth all federal tax elections under the Internal Revenue Code of 1986 (the "Code") filed since January 1, 1996 that are in effect with respect to the Nationwide Entities or for which an application by any of the Nationwide Entities is pending.
- (b) There are no levies or liens on any of the assets of any Nationwide Entity that arose in connection with any failure (or alleged failure) to pay Taxes (as defined in subsection 2.8(e) below) when due. No Nationwide Entity has ever entered into a closing agreement pursuant to Section 7121 of the Internal Revenue Code.

15

- (c) Except as set forth in <u>Schedule 2.8</u>, there has not been any audit of any tax return filed by the Nationwide Entities since January 1, 1996, no audit of any federal or state tax return of the Nationwide Entities is in progress, and none of the Nationwide Entities or the Shareholders has been notified by any federal or state tax authority that any such audit is contemplated or pending. Except as set forth in <u>Schedule 2.8</u>, no extension of time with respect to any date on which a federal or state tax return was or is to be filed by the Nationwide Entities is in force, and no waiver or agreement by the Nationwide Entities is in force for the extension of time for the assessment or payment of any Taxes. Since January 1, 1996, none of the Nationwide Entities has had a federal or state tax deficiency proposed, asserted or assessed against it, or has been delinquent in the payment or deposit of any Taxes.
- (d) Except as set forth in <u>Schedule 2.8</u> since January 1, 1996, all federal, state and local income tax and information returns have been filed, are true and correct in all material respects and any and all Taxes shown as due and owing thereon have been timely and fully paid, and any and all Taxes required to be withheld and deposited have been withheld and deposited.
- (e) For purposes of this Agreement, the term "Taxes" shall mean all federal, state, and local taxes, including, without limitation, income taxes, estimated taxes, alternative minimum taxes, excise taxes, sales taxes, use taxes, value-added taxes, gross receipts taxes, franchise taxes, capital stock taxes, employment and payroll-related taxes, withholding taxes, stamp taxes, transfer taxes and property taxes, whether or not measured in whole or in part by net income, and all deficiencies, or other additions to tax, interest, fines and penalties.
- 2.9 <u>Accounts Receivable</u>. All accounts receivable of the Nationwide Entities reflected on the Audited Balance Sheet and the Base Balance Sheet arise out of transactions in the ordinary course of business and any reserves established therefor are calculated consistent with past practices. The Nationwide Entities have no accounts or loans receivable from the Shareholders, any person, firm or corporation which is affiliated with the Shareholders or the Nationwide Entities or from any affiliate, director, officer or employee of the Shareholders or the Nationwide Entities, except as disclosed on <u>Schedule 2.9</u> attached hereto.
- 2.10 <u>Inventories</u>. The values of the inventories stated in the Audited Balance Sheet and the Base Balance Sheet were determined in accordance with GAAP subject to normal year-end audit adjustments.
- 2.11 <u>Absence of Certain Changes</u>. Except as disclosed in <u>Schedule 2:11</u> attached hereto or any other Schedule attached hereto, since the date of the Base Balance Sheet there has not been:
  - (i) Any material change in the properties, assets, liabilities, business, operations or condition (financial or other) of any of the Nationwide Entities, except those arising in the ordinary course of business;

16

- (ii) Except for the endorsement of checks in the ordinary course of business, any contingent liability incurred by any of the Nationwide Entities as guarantor;
- (iii) Any mortgage, encumbrance or lien for borrowed money placed on any of the properties of either of the Nationwide Entities which will remain in existence after the Closing Date;
- (iv) Any purchase, sale or other disposition, or any agreement or other arrangement for the purchase, sale or other disposition, of any individual capital asset of either of the Nationwide Entities costing more than \$10,000;
- (v) Any damage, destruction or loss, whether or not covered by insurance, suffered by either of the Nationwide Entities involving an amount in excess of \$10,000;
- (vi) Any declaration, setting aside or payment of any dividend by any of the Nationwide Entities, or the making of any other distribution in respect of the equity of the Nationwide Entities (except as has been consistent with past practices or is intended to result in the Nationwide Entities having the 1998 Agreed Equity), any changes in the capitalization of either of the Nationwide Entities, any direct or indirect redemption, purchase or other acquisition by either of the Nationwide Entities of its own capital stock, any issuance or sale of any securities convertible into or exchangeable for debt or equity securities of the Nationwide Entities or any grant or issuance of options, warrants, subscriptions, preemptive rights, agreements, arrangements or commitments of any kind for or relating to the issuance, sale, registration or voting of any shares of capital stock of any class, or other equity interests of any of the Nationwide Entities, as applicable;
- (vii) Any claim of unfair labor practices asserted against any of the Nationwide Entities; any change in the compensation (in the form of salaries, wages, incentive arrangements or otherwise) payable or to become payable by the Nationwide Entities to any of its officers, employees or John Folz or Alan Carlitz other than normal merit increases in accordance with its usual practices, or any bonus payment or arrangement made to or with any of such officers, employees or such named independent contractors; any entering into any employment, deferred compensation or other similar agreement (or any amendment to any such existing agreement) with any officer, director or employee of the Nationwide Entities, except for any employment arrangement providing for salary or wages of less than \$40,000 per annum and any oral agreement terminable at will by the Nationwide Entities;
- (viii) Any change in the officers or senior management of either of the Nationwide Entities, any grant of or any agreement to grant any severance or

17

termination pay to any officer or employee of the Nationwide Entities or any increase in benefits payable under any existing severance or termination pay policies or employment agreements except in accordance with past practices (which is not to exceed one week for each year of service);

- (ix) Any obligation or liability incurred by either of the Nationwide Entities to any of its officers, directors, stockholders or employees which exceed, in each instance, \$2,000 per person in the aggregate, or any loans or advances made by any of the Nationwide Entities to any of its officers, directors, stockholders or employees, which exceed, in each instance, \$2,000 per person in the aggregate except compensation and expense allowances payable to officers or employees;
- (x) Any change in accounting methods or practices, credit practices or collection policies used by the Nationwide Entities;
- (xi) Any other transaction entered into by either of the Nationwide Entities other than transactions in the ordinary course of business or as contemplated by this Agreement; or
- (xii) Any agreement or understanding whether in writing or otherwise, that would result in any of the transactions or events or require any of the Nationwide Entities to take any of the actions specified in paragraphs (i) through (xi) above.
- 2.12 <u>Banking Relations</u>. All of the accounts which the Nationwide Entities have with any banking institution are listed in <u>Schedule 2.12</u> attached hereto, indicating with respect to each of such accounts the type of accounts maintained (such as checking account, borrowing arrangements, safe deposit box, etc.), the name of the appropriate Nationwide Entity with whom such accounts have been made and the person or persons authorized in respect thereof.
- 2.13 Patents, Tradenames, Trademarks, Copyrights and Proprietary Rights. All patents, patent applications, trade names, trademarks, trademark applications and registrations, copyrights, applications and registrations and other material proprietary rights owned by or licensed to the Nationwide Entities or used or to be used by either of the Nationwide Entities in the Business as presently conducted (the "Proprietary Rights") are listed in Schedule 2.13 attached hereto. Except as set forth in Schedule 2.13, to their knowledge, use of said patents, trade names, trademarks, copyrights or other proprietary rights in the ordinary course of the business as presently conducted does not require the consent of any other person. Except as set forth in Schedule 2.13, to their knowledge, the Nationwide Entities have sufficient title or adequate rights or licenses to use of all patent, tradename, trademark, copyright, or other proprietary rights used or to be used by the Nationwide Entities in their Business as presently conducted or contemplated, in each case free and clear of any attachments, liens, encumbrances or adverse claims and the Nationwide Entities have no reason to believe that its present or contemplated activities, products or services infringe any such

18

patent, trade name, trademark or other proprietary rights of others. Except as set forth in <u>Schedule 2.13</u>, (a) none of the patents, trade names, trademarks, copyrights or other proprietary rights listed in <u>Schedule 2.13</u> is subject to any outstanding order, decree, judgment or stipulation or, to the best knowledge of the Nationwide Entities and the Shareholders, is being infringed by others, and (b) no proceeding charging the Nationwide Entities, or any of them, with infringement of any adversely held patent, trade name, trademark or copyright has been filed or, to the best knowledge of the Nationwide Entities and the Shareholders, is threatened to be filed, and neither the Nationwide Entities nor the Shareholders has received notice of any such proceeding.

- 2.14 <u>Trade Secrets and Customer Lists</u>. The Nationwide Entities have the right to use in the ordinary course of their Business as presently conducted all trade secrets and customer lists required for or incident to the marketing of all products presently sold by the Nationwide Entities. Any payments required to be made by the Nationwide Entities for the use of such trade secrets and customer lists are described in <u>Schedule 2.14</u>. The Nationwide Entities are not using or in any way making use of any confidential information or trade secrets of any third party, including without limitation, a former employer of any present or past employee of the Nationwide Entities or of any of the Nationwide Entities' predecessors, except where such use or the loss of the same would not have a material adverse effect on the properties, assets, business or condition (financial or other) of the Nationwide Entities or the Business.
- 2.15 <u>Contracts</u>. Except for contracts, commitments, plans, agreements and licenses listed in <u>Schedule 2.15</u> (true and complete copies of those of which are in writing have been delivered to the Parent), none of the Nationwide Entities is currently a party to or subject to:
- (a) any plan or contract providing for bonuses, pensions, options, stock purchases, deferred compensation, retirement payments, profit sharing, collective bargaining or the like, or any contract or agreement with any labor union;
- (b) any employment contract or contract for services which requires the payment of more than \$40,000 annually in base salary or which is not terminable within 30 days by the Nationwide Entities without liability for any penalty or severance payment other than pursuant to the Nationwide Entities' severance policies referred to on Schedule 2.24;
- (c) any contract or agreement for the purchase of any commodity, material or equipment other than those purchased in the ordinary course of business;
- (d) except for the agreements disclosed in any schedule hereto, any other contracts or agreements creating any obligation of the Nationwide Entities of \$50,000 or more with respect to any such contract;
- (e) any contract or agreement which by its terms does not terminate or is not terminable by the Nationwide Entities or any successor or assign within twelve months after the date hereof without payment of a penalty of \$10,000 or more;

19

- (f) any contract or agreement for the sale or lease of its products not made in the ordinary course of business;
- (g) any contract with any sales agent or distributor of products of either of the Nationwide Entities;
- (h) any contract or agreement containing covenants limiting the freedom of any of the Nationwide Entities to compete in any line of business or territory or with any person or entity;
- (i) any contract or agreement which restricts the conduct of the Business in any form or fashion which contract or agreement will survive Closing and is not terminable at will;
- (j) except for obligations incurred for any new store openings from and after December 31, 1997 (each of which is listed on Schedule 2.15), any outstanding contract or agreement for the purchase of any individual fixed asset for a price in excess of \$10,000 whether or not such purchase is in the ordinary course of business;
- (k) any license agreement (as licensor or licensee) which is material to the Business;
- (l) any indenture, mortgage, promissory note, loan agreement, guaranty or other agreement or commitment for the borrowing of money and any related security agreement;
- (m) any partnership, joint venture or other similar contract, arrangement or agreement;
- (n) except as included in any leases disclosed to the Parent pursuant to Section 2.6 hereof, any contract providing for indemnification of liabilities under Environmental Laws (as defined in Section 2.26);
- (o) any guaranty, suretyship agreement or other contract or agreement relating to any contingent liability or obligations of any third party; or
  - (p) any power of attorney, whether general, special or limited, or proxy.

All contracts, agreements, leases and instruments to which any of the Nationwide Entities is a party or by which either of the Nationwide Entities is obligated are legal, valid and binding obligations of the respective Nationwide Entity and, to the best knowledge of the Nationwide Entities and the Shareholders, the other parties thereto, enforceable in accordance with their respective terms. Neither the Nationwide Entities nor, to the best knowledge of the Nationwide Entities and the Shareholders, any other party to any contract, agreement, lease or instrument of any of the Nationwide Entities including, without limiting the generality of the foregoing, any agreement relating to continuing warranty or service obligations, is in material default in complying with any

20

provisions thereof, and no condition or event or facts exist which, with notice, lapse of time or both, would constitute a default thereof on the part of such Nationwide Entity or, to the best knowledge of the Nationwide Entities and the Shareholders, on the part of any other party thereto, except for any such default, condition, event or facts that, individually or in the aggregate, would not have a material adverse effect on the properties, assets, business, condition (financial or other) or prospects of the Nationwide Entities or the Business.

- 2.16 <u>Litigation</u>. <u>Schedule 2.16</u> hereto lists all currently pending litigation and governmental or administrative proceedings or, to their knowledge, investigations to which any of the Nationwide Entities is a party. Except for matters described in <u>Schedule 2.16</u>, there is no litigation or governmental or administrative proceeding or investigation pending or, to their knowledge, threatened against any of the Nationwide Entities which may have an adverse effect on the properties, assets, condition (financial or other) of the Nationwide Entities or the Business or which would prevent or hinder the consummation of the transactions pursuant to, or as contemplated by, this Agreement.
- 2.17 <u>Compliance with Laws</u>. Except as set forth in <u>Schedule 2.17</u> hereto, to their knowledge, each of the Nationwide Entities is in compliance with all applicable statutes, ordinances, orders, rules and regulations promulgated by any federal, state, municipal or other governmental authority which apply to the conduct of the Business, except for any such non-compliance or violation that, individually or in the aggregate, would not have an adverse effect on the properties, assets, business or condition (financial or other) of the Nationwide Entities or the Business, and none of the Nationwide Entities has received any written notice of a violation or alleged violation of any such statute, ordinance, order, rule or regulation.
- 2.18 <u>Insurance</u>. The physical properties, assets, business, operations, employees, officers and directors of the Nationwide Entities are insured to the extent disclosed in <u>Schedule 2.18</u> attached hereto and all insurance policies and arrangements of the Nationwide Entities are disclosed in said Schedule. Neither the execution of this Agreement nor the consummation of the transactions contemplated hereby will cause any of such insurance policies or arrangements by their terms to lapse or be cancelled or result in any modification under any such policies or arrangements including any increase in premiums. Said insurance policies and arrangements are in full force and effect and all premiums now due with respect thereto are currently paid. There is no outstanding claim by any of the Nationwide Entities pending under any such policies as to which coverage has been denied or disputed by the insurer, except as described on <u>Schedule 2.18</u>.
- 2.19 <u>Finder's Fees</u>. Neither the Nationwide Entities nor any Shareholder has incurred or become liable for any broker's commission or finder's fee relating to or in connection with the transactions contemplated by this Agreement other than as set forth on <u>Schedule 2.19</u> attached hereto.
- 2.20 <u>Permits</u>. <u>Schedule 2.20</u> attached hereto lists all of the Nationwide Entities' material permits, registrations, licenses, franchises, certifications and other approvals (collectively, the

21

"Approvals") from federal, state or local authorities with respect to the conduct of the Business. Each of the Nationwide Entities has obtained all Approvals necessary for the conduct of the Business as presently conducted, and all such Approvals are valid and in full force and effect, except where the lack of such Approval would not have an adverse effect on the properties, assets, business or condition (financial or other) of the Nationwide Entities or on the Business. Except as disclosed in Schedule 2.20, none of such Approvals is subject to termination by its express terms as a result of the execution of this Agreement by the Nationwide Entities or the consummation of the transactions contemplated hereby, and no further Approvals will be required in order to continue to conduct the Business as currently conducted by the Nationwide Entities subsequent to the Closing and consummation of the Merger Transactions, except where the termination of such Approvals or the lack of such further Approvals would not have an adverse effect on the properties, assets, business or condition (financial or other) of the Nationwide Entities.

2.21 <u>Transactions with Interested Persons</u>. Except as set forth in <u>Schedule 2.21</u> hereto, none of the Nationwide Entities, any Shareholder, any officer or director of the Nationwide Entities nor any of their respective spouses or children or trusts established for such spouses or children owns directly or indirectly on an individual or joint basis any material interest in, or serves as an officer or director or in another similar capacity of, any competitor or supplier of any of the Nationwide Entities, or any organization which has a material contract or arrangement with any of the Nationwide Entities.

# 2.22 Employee Benefit Programs.

- (a) Schedule 2.22 attached hereto sets forth a list of every Employee Program (as defined in subsection (f)) that has been maintained (as such term is further defined below) by any of the Nationwide Entities at any time since January 1, 1998.
- (b) Except as described in Schedule 2.22, each Employee Program which has been maintained by any of the Nationwide Entities and which has at any time been intended to qualify under Code Section 401(a) or 501(c)(9) of the Code has received a favorable determination or approval letter from the Internal Revenue Service ("IRS") regarding its qualification under such section and such favorable determination or approval letter remains in force. In the case of any Employee Program maintained by any of the Nationwide Entities and intended to qualify under Code Section 401(a) or 501(c)(9) which is described in Schedule 2.22 as not having received a favorable determination letter from the IRS, no circumstances exist that would cause such Employee Program to fail to qualify under Code Section 401(a) or 501(c)(9), whichever is relevant. All required reports and descriptions, including without limitation annual reports (Form 5500), summary annual reports and summary plan descriptions, have been filed and distributed.
- (c) There has not been any failure of either of the Nationwide Entities to substantially comply with any laws applicable with respect to the Employee Programs that have been maintained by any of the Nationwide Entities, except for any failures to comply that, individually or in the aggregate, would not have a material adverse effect on the properties, assets, business or

22

condition (financial or other) of the Nationwide Entities. With respect to any Employee Program now or heretofore maintained by any of the Nationwide Entities, there has occurred no "prohibited transaction," as defined in Section 406 and 408 of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), or Section 4975 of the Code since January 1, 1998. No litigation, arbitration, or governmental administrative proceeding (or investigation) or other proceeding (other than those relating to routine claims for benefits) is pending or, to the best knowledge of the Nationwide Entities or the Shareholders, threatened with respect to any such Employee Program, and neither the Nationwide Entities nor the Shareholders have knowledge of any facts that could form the basis for any such litigation, arbitration or governmental administrative proceeding or investigation. Neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated by this Agreement will entitle any person to any additional or other benefits, or will otherwise modify benefits or the vesting of benefits, under any Employee Program.

- (d) Neither the Nationwide Entities nor any Affiliate (as defined in subsection (f)) has incurred any outstanding liability (other than relating to premiums timely paid to the Pension Benefit Guaranty Corporation in the ordinary course) under title IV of ERISA and there currently is no "accumulated funding deficiency" (whether or not waived) with respect to any Employee Program currently maintained by either of the Nationwide Entities or any Affiliate and subject to Code Section 412 or ERISA Section 302. With respect to any Employee Program maintained by any of the Nationwide Entities or an Affiliate and subject to title IV of ERISA, there currently is no (i) "reportable event," within the meaning of ERISA Section 4043, or the regulations thereunder (for which notice the notice requirement is not waived under 29 C.F.R. Part 2615) or (ii) event or condition which will cause any of the Nationwide Entities or any Affiliate to incur liability or have a lien imposed on its assets under title IV of ERISA. Except as described in Schedule 2.22 no Employee Program maintained by either of the Nationwide Entities or an Affiliate and subject to title IV of ERISA (other than a Multiemployer Plan) has any "unfunded benefit liabilities" within the meaning of ERISA Section 4001(a)(18), as of the Closing Date. None of the Nationwide Entities nor any Affiliate has ever maintained a Multiemployer Plan since the date of incorporation of such Nationwide Entities except with each other.
- (e) With respect to each Employee Program currently maintained by either of the Nationwide Entities since the date of incorporation of each, complete and correct copies of the following documents (if applicable to such Employee Program) have previously been delivered to the Parent, except as set forth in Schedule 2.22: (i) all documents embodying or governing such Employee Program, and any funding medium for the Employee Program (including, without limitation, trust agreements) as they may have been amended to the date hereof; (ii) the most recent IRS determination or approval letter with respect to such Employee Program under Code Section 401 or 501(c)(9), and any applications for determination or approval subsequently filed with the IRS; (iii) the three most recently filed IRS Forms 5500, with all applicable schedules and accountants' opinions attached thereto; (iv) the summary plan description for such Employee Program and all modifications thereto; and (v) any insurance policy (including any fiduciary liability insurance policy) related to such Employee Program.

23

# (f) For purposes of this <u>Section 2.22</u>:

- (i) "Employee Program" means (A) all employee benefit plans within the meaning of ERISA Section 3(3), including, but not limited to, multiple employer welfare arrangements (within the meaning of ERISA Section 3(40)), plans to which more than one unaffiliated employer contributes and employee benefit plans (such as foreign or excess benefit plans) which are not subject to ERISA; and (B) all stock option plans, bonus or incentive award plans, severance pay policies or agreements, deferred compensation agreements, supplemental income arrangements, vacation plans, and all other employee benefit plans, agreements and arrangements not described in (A) above, but excluding base compensation and annual discretionary cash bonuses. In the case of an Employee Program funded through an organization described in Code Section 501(c)(9), each reference to such Employee Program shall include a reference to such organization;
- (ii) An entity "maintains" an Employee Program if such entity sponsors, contributes to, or provides (or has promised to provide) benefits under such Employee Program, or has any obligation (by agreement or under applicable law) to contribute to or provide benefits under such Employee Program, or if such Employee Program provides benefits to or otherwise covers employees of such entity (or their spouses, dependents, or beneficiaries);
- (iii) An entity is an "Affiliate" of the Nationwide Entities for purposes of this Section 2.22 if it would have ever been considered a single employer with any of the Nationwide Entities under ERISA Section 4001(b) or part of the same "controlled group" as the Nationwide Entities for purposes of ERISA Section 302(d)(8)(C); and
- (iv) "Multiemployer Plan" means a (pension or non-pension) employee benefit plan to which more than one employer contributes and which is maintained pursuant to one or more collective bargaining agreements.

# 2.23 <u>List of Certain Employees and Suppliers</u>.

(a) Schedule 2.23(a) attached hereto contains a list of all current directors and officers of the Nationwide Entities and a list of all managers, employees and consultants of the Nationwide Entities who, individually, have received or are scheduled to receive from the Nationwide Entities for the fiscal year of the Nationwide Entities ending December 31, 1998 base salary and/or other compensation in excess of \$40,000. In each case such Schedule includes, other than for the Shareholders, the current job title and current base salary or compensation structure of each such individual.

- (b) <u>Schedule 2.23(b)</u> attached hereto sets forth a true and complete list of all suppliers (each, a "Supplier") of the Nationwide Entities to whom during the nine months ended on the date of the Base Balance Sheet, the Nationwide Entities made payments aggregating \$100,000 or more showing, with respect to each, the name and dollar volume involved. Since the date of the Base Balance Sheet, no Supplier has given written or oral notice to any of the Shareholders or the Nationwide Entities of its intent to terminate, cancel or otherwise materially and adversely modify its relationship with the Nationwide Entities or to limit or decrease its services, products, supplies or materials to any of the Nationwide Entities.
- 2.24 Employees: Labor Matters. None of the Nationwide Entities is delinquent in payments to any of its employees for any wages, salaries, commissions, bonuses or other direct compensation for any services performed for it to the date hereof or amounts required to be reimbursed to such employees. Except as set forth in Schedule 2.24 attached hereto, none of the Nationwide Entities has any policy, practice, plan or program of paying severance pay or any form of severance compensation in connection with the termination of employment. Each of the Nationwide Entities is in compliance with all applicable laws and regulations respecting labor, employment, fair employment practices, terms and conditions of employment, and wages and hours, except for any non-compliance that, individually or in the aggregate, would not have an adverse effect on the properties, assets, business or condition (financial or other) of the Nationwide Entities. No charges of employment discrimination or unfair labor practices are currently being brought against either of the Nationwide Entities or the Shareholders that might have an adverse affect on the Nationwide Entities or the conduct of its Business, nor are there any strikes, slowdowns, stoppages of work, or any other concerted interference with normal operations existing, or to their knowledge, threatened against or involving either of the Nationwide Entities. None of the Nationwide Entities has received written notice of any impending strikes, slowdowns, concerted interference with normal operations or union organization activities. Except as set forth in Schedule 2.24, there are no grievances, complaints or charges that have been filed against any of the Nationwide Entities under any dispute resolution procedure (including, but not limited to, any proceedings under any dispute resolution procedure under any collective bargaining agreement) that might have an adverse effect on the Nationwide Entities or the conduct of its Business. Except as set forth in Schedule 2.24, no collective bargaining agreements are in effect or are currently being or are about to be negotiated by any of the Nationwide Entities. None of the Nationwide Entities has received written notice of pending or threatened charges with respect to (including, without limitation, resignation of) the senior management or key supervisory personnel of the Nationwide Entities.
- 2.25 <u>Customers</u> None of the Nationwide Entities have any customers who purchase product or services in an aggregate amount in excess of \$25,000 in any calendar year.
  - 2.26 Environmental Matters. Except as set forth on Schedule 2.26:
- (a) no Hazardous Material (i) to their knowledge has been released, placed, stored, generated, used, manufactured, treated, deposited, spilled, discharged, released or disposed

of, on or under any real property currently or previously owned or leased by any of the Nationwide Entities, (ii) is presently maintained, used, generated, or permitted to remain in place by any of the Nationwide Entities in violation of any Environmental Law, (iii) to their knowledge, is required by any Environmental Law to be eliminated, removed, treated or mitigated by any of the Nationwide Entities, given the nature of its present condition, location, nature, material or maintenance, or (iv) to their knowledge, is of a type, location, material, nature or condition which requires special notification to third parties by any of the Nationwide Entities under any Environmental Law or common law;

- (b) Since January 1, 1997, no outstanding notice, citation, summons, or order has been received by any of the Nationwide Entities, no notice has been given by any of the Nationwide Entities and no complaint has been filed, no penalty has been assessed and no investigation or review is pending or, to their knowledge, threatened by any governmental entity, with respect to (i) any alleged violation by any of the Nationwide Entities of any Environmental Law, or (ii) any alleged failure by any of the Nationwide Entities to have any environmental permit, certificate, license, approval, registration or authorization required in connection with its Business or properties in violation of an Environmental Law, or (iii) any use, possession, generation, treatment, storage, recycling, transportation, release or disposal by or on behalf of any of the Nationwide Entities of any Hazardous Material in violation of an Environmental Law;
- (c) Since January 1, 1997 neither the Nationwide Entities nor any Shareholder has received any written request for information, notice of claim, demand or notification that it is or that indicates that it may be a "potentially responsible party" with respect to any investigation or remediation of any threatened or actual release of any Hazardous Material:
- (d) Since January 1, 1997 no written notice has been received by the Nationwide Entities with respect to the listing or proposed listing of any property currently or previously owned, operated or leased by any of the Nationwide Entities on the National Priorities List promulgated pursuant to CERCLA, CERCLIS or any similar state list of sites requiring investigation or cleanup; and
- (e) neither of the Nationwide Entities has released, transported, or arranged for the transportation of any Hazardous Material from any property currently or previously owned, operated or leased by the Nationwide Entities in violation of any Environmental Law.

For purposes of this <u>Section 2.26</u>, "Environmental Law" shall mean any environmental law, regulation, rule or ordinance, at the federal, state, or local level and "Hazardous Material" shall mean any flammable, ignitable, corrosive, reactive, radioactive or explosive substance or material, hazardous waste, toxic substance or related material and any other substance or material defined or designated as a hazardous or toxic substance, material or waste by any Environmental Law other than typical cleaning supplies and solvents.

26

2.27 <u>Disclosure</u>. The representations, warranties and statements contained in this Agreement and in the certificates, exhibits and schedules delivered by the Nationwide Entities or the Shareholders to the Parent pursuant to this Agreement contain all of the representations and warranties made by the Nationwide Entities and the Shareholders to the Parent and do not contain any untrue statement of a material fact.

# ARTICLE 3. COVENANTS OF THE NATIONWIDE ENTITIES AND THE SHAREHOLDERS.

- 3.1 <u>Making of Covenants and Agreements</u>. The Nationwide Entities and the Shareholders jointly and severally (except as noted herein) make the covenants and agreements as set forth in this Article 3 which covenants and agreements are applicable from this date to the Closing Date.
- 3.2 <u>Conduct of Business</u>. Except as set forth in <u>Schedule 3.2</u>, between the date of this Agreement and the Closing Date, the Shareholders will cause the Nationwide Entities to do and the Nationwide Entities will do the following, unless the Parent shall otherwise consent in writing:
- (a) conduct the Business only in the ordinary course consistent with past practices;
- (b) refrain from making any purchase, sale or disposition of any asset or property other than in the ordinary course of business or as contemplated herein, and from purchasing or selling any capital asset costing more than \$10,000;
- (c) refrain from incurring or modifying any contingent liability as a guarantor with respect to the obligations of others, and from incurring or modifying any other contingent or fixed obligations or liabilities, other than as provided herein, except in the ordinary course of business and consistent with past practices;
- (d) refrain from making any change in its organizational or incorporation documents, by laws or authorized or issued capital stock and refrain from any merger or consolidation with or into any other person or entity;
- (e) refrain from (i) declaring, setting aside or paying any dividend (except as contemplated in Section 2.11 hereof), (ii) making any other distribution in respect of its capital stock, making any direct or indirect redemption, purchase or other acquisition of its capital stock, (iii) issuing, granting, awarding, selling, pledging, disposing of or encumbering or authorizing the issuance, grant, award, sale, pledge, disposition or encumbrance of any shares of, or securities convertible or exchangeable for, or options, warrants, calls, commitments or rights of any kind to acquire, any shares of capital stock of any class of the Nationwide Entities and/or (iv) entering into any agreement or commitment with respect to any of the foregoing;

27

- (f) refrain from (i) making any payments to its Shareholders except as set forth in Schedule 3.2(f), (ii) making any change in the compensation payable or to become payable to any of its officers, employees or agents, except for scheduled increases in salary or wages in the ordinary course of business consistent with past practices or increases or changes which do not survive the Closing Date, (iii) granting any severance or termination pay (except in the ordinary course of business) to, or entering into or amending any employment, severance or other agreement or arrangement with, any director, officer or other employee of any of the Nationwide Entities, and/or (iv) establishing, adopting or entering into or amending any collective bargaining, bonus, incentive, deferred compensation, profit sharing, stock option or purchase, insurance, pension, retirement or other employee benefit plan;
- (g) use its good faith efforts, consistent with this Agreement, to prevent any change with respect to its management and supervisory personnel or banking arrangements;
- (h) use its good faith efforts, consistent with this Agreement, to keep intact its business organization and to preserve the goodwill of and business relationships with all suppliers, customers and others having business relations with it;
- (i) pay all accounts payable in the ordinary course of business consistent with past practices unless such accounts payable are being disputed in good faith;
- (j) use its good faith efforts to have in effect and maintain at all times all insurance of the kind, in the amount and with the insurers set forth in <u>Schedule 2.18</u> or equivalent insurance;
- (k) refrain from changing accounting policies or procedures or from making any tax election except where such election is made in connection with the filing of federal or state tax returns and would not have an adverse effect on the properties, assets, business or condition (financial or other) of any of the Nationwide Entities or the Business;
- (l) refrain from entering into any executory agreement, commitment or undertaking to do any of the activities prohibited by the foregoing provisions, unless the effectiveness of such agreement, commitment or undertaking is expressly conditioned on the termination of this Agreement pursuant to its terms; and
- (m) permit the Parent and its authorized representatives (including without limitation the Parent's attorneys, accountants, investment bankers and lenders) to have reasonable access during normal business hours at the offices of the Nationwide Entities to all of its properties, assets, books, records, business files, executive personnel, tax returns, contracts and documents except the information referenced in <u>Schedule 3.2(n)</u> and furnish to the Parent and its authorized representatives such financial and other information with respect to its business or properties as the Parent may from time to time reasonably request and which is reasonably available to the Nationwide Entities.

28

Notwithstanding the foregoing, the parties agree and acknowledge that the Nationwide Entities intend to and shall have the right to sell, convey, assign, divide, distribute or otherwise transfer, at no cost to the Shareholders, to the Shareholders these assets of the Nationwide Entities set forth in Schedule 3.2 ("Excluded Assets").

- 3.3 <u>Consents and Approvals</u>. The Nationwide Entities and each of the Shareholders shall use their commercially reasonable efforts to obtain or cause to be obtained prior to the Closing Date all necessary consents and approvals to the performance of its obligations under this Agreement, including, without limitation, such authorizations, waivers, consents and permits as may be necessary to retain in full force and effect subsequent to the Closing Date all permits, licenses, agreements and franchises applicable to the Nationwide Entities.
- 3.4 <u>Breach of Representations and Warranties</u>. Promptly upon the occurrence of, or promptly upon any of the Nationwide Entities or any Shareholder becoming aware of, the impending or threatened occurrence of any event which would cause or constitute a breach, or would have caused or constituted a breach had such event occurred or been known prior to the date hereof, of any of the representations and warranties of any of the Nationwide Entities or the Shareholders contained in or referred to in this Agreement or in any Schedule hereto, such Nationwide Entity or such Shareholder shall give detailed written notice thereof to the Parent.
- 3.5 <u>Acquisition Proposals</u>. Unless and until this Agreement shall have been terminated pursuant to Article 7, neither the Nationwide Entities nor any Shareholder shall solicit or initiate submission of proposals or offers from any person relating to any acquisition or purchase of all or (other than in the ordinary course of business) a material portion of the assets of, or any equity interest in, the Nationwide Entities or any merger or business combination with the Nationwide Entities (an "Acquisition Proposal") or participate in any discussions or negotiations regarding an Acquisition Proposal, or furnish to any other person any information with respect to or afford access to the properties, books or records of the Nationwide Entities to any person that may consider making or has made an offer with respect to an Acquisition Proposal, or otherwise cooperate in any way with, or assist or participate in or facilitate any effort or attempt by any other person to do or seek any of the foregoing. A breach by an individual Shareholder of the covenants contained in this Section shall not be considered a breach by either of the Nationwide Entities or any other Shareholder, and such other Shareholders and the Nationwide Entities shall not incur any liability therefor.
- 3.6 <u>No Sales of Nationwide Entities Equity</u>. Between the date of this Agreement and the Closing, none of the Shareholders shall sell, exchange, deliver, assign, pledge, encumber, bequeath, gift or otherwise transfer or dispose of any shares of capital stock (or any interest therein) of any of the Nationwide Entities owned beneficially or of record by such Shareholder, except for any transfer or other disposition by operation of law, nor grant any right of any kind to acquire, dispose of, vote or otherwise control in any manner such shares of capital stock of any of the Nationwide Entities; provided, however, that anything herein to the contrary notwithstanding, any transferee, executor, heir, legal representative, successor or assign of any Shareholder shall be bound by this Agreement.

29

A breach by an individual Shareholder of the covenants contained in this Section shall not be considered a breach by either of the Nationwide Entities or any other Shareholder, and such other Shareholders and the Nationwide Entities shall not incur any liability therefor.

- Shareholders shall use their good faith efforts, in addition to the performance and fulfillment of all covenants, agreements, conditions and obligations on their parts to be performed and fulfilled under this Agreement, to the end that the Merger Transactions contemplated by this Agreement shall be fully carried by the Closing Date, and shall cooperate toward such end with all reasonable requests of the Parent and the Parent's counsel in connection with the consummation of the Merger Transactions contemplated hereby including the provision of all current financial information regarding the Business as may be required or requested by the Parent or its senior lender to allow the Parent to obtain the debt financing sufficient to pay the Purchase Price and all of the Parent's closing costs, fees and expenses. Furthermore, the Nationwide Entities will allow one or more representatives of the Parent access to the facilities of the Nationwide Entities, subject to the Nationwide Entities' reasonable operational requirements and procedures.
- 3.8 <u>Termination of 401(K) Plan</u>. The Nationwide Entities intend to terminate their existing 401(K) plans relating to their employees prior to the Closing Date.

## ARTICLE 4. REPRESENTATIONS AND WARRANTIES OF MGHC AND PARENT.

- 4.1 <u>Making of Representations and Warranties</u>. Each of MGHC, Parent and the MG Merger Entities hereby jointly and severally represents and warrants to the Shareholders and the Nationwide Entities as follows:
- 4.2 <u>Organization</u>. Each of MGHC and Parent is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Each of the MG Merger Entities is a corporation duly organized, validly existing and in good standing under the laws of the Commonwealth of Pennsylvania.
- Authority. All necessary corporate action has been taken by each of MGHC, the Parent and the MG Merger Entities to authorize the execution, delivery and performance of this Agreement and each agreement, document and instrument to be executed and delivered by MGHC, the Parent or the MG Merger Entities pursuant to this Agreement. This Agreement and each agreement, document and instrument to be executed and delivered by MGHC, the Parent or the MG Merger Entities pursuant to this Agreement (to the extent it contains obligations to be performed by MGHC or the Parent) constitutes, or when executed and delivered will constitute, valid and binding obligations of the MGHC, Parent or the MG Merger Entities, as appropriate, enforceable in accordance with their respective terms. The execution, delivery and performance by MGHC, the Parent and the MG Merger Entities of this Agreement and each such agreement, document and instrument:

30

- (a) do not and will not violate any provisions of the Certificate of Incorporation or By-Laws of MGHC, the Parent and the MG Merger Entities, each as amended through the date hereof:
- (b) do not and will not violate any United States laws or laws of the jurisdiction of incorporation of MGHC, the Parent and the MG Merger Entities or of any other jurisdiction, or require MGHC, the Parent or the MG Merger Entities to obtain any approval, consent or waiver of, or to make any filing with, any person or entity that has not been obtained or made or will not be obtained or made prior to the Closing Date; and
- (c) do not and will not result in a breach of, constitute a default under, accelerate any obligation under or give rise to a right of termination of any indenture or loan or credit agreement or any other agreement, contract, instrument, mortgage, lien, order, writ, judgment, injunction, decree, determination or arbitration award to which MGHC, Parent or the MG Merger Entities is a party or by which the property of MGHC, the Parent or the MG Merger Entities is bound or affected except where MGHC, the Parent or the MG Merger Entities shall obtain a waiver of such breach, default, acceleration or right of termination prior to the Closing Date.
- 4.4 <u>Litigation</u>. There is no litigation pending or, to the best knowledge of the MGHC, Parent and the MG Merger Entities, threatened against MGHC, the Parent or the MG Merger Entities which would prevent or hinder the consummation of the transactions contemplated by this Agreement.
- 4.5 <u>Equity</u>. All of the issued and outstanding capital stock of the Parent is owned beneficially and of record by MGHC. All of the issued and outstanding capital stock of the MG Merger Entities is owned beneficially and of record by the Parent.
- 4.6 <u>Compliance with Laws</u>. Each of MGHC, the Parent and the MG Merger Entities is in compliance with all applicable statutes, ordinances, orders, rules and regulations promulgated by any federal, state, municipal or other governmental authority which apply to the conduct of their business, except for any such non-compliance or violation that, individually or in the aggregate, would not have an adverse effect on their properties, assets, business or condition (financial or other), and none of MGHC, the Parent and the MG Merger Entities has received any written notice of a violation or alleged violation of any such statute, ordinance, order, rule or regulation.
- 4.7 <u>Finder's Fees</u>. Neither MGHC nor Parent has incurred or become liable for any broker's commission or finder's fee relating to or in connection with this Agreement or the transactions contemplated hereby.
- 4.8 <u>Disclosure</u>. The representations, warranties and statements contained in this Agreement and in the certificates, exhibits and schedules delivered by MGHC, the Parent and the MG Merger Entities to the Nationwide Entities and the Shareholders pursuant to this Agreement do not contain any untrue statement of a material fact.

31

4.9 <u>Survival</u>. All representations and warranties made in this Article 4 shall survive for a period of twelve (12) months from the Closing Date.

## ARTICLE 5. COVENANTS OF MGHC AND PARENT.

- 5.1 <u>Making of Covenants and Agreements</u>. MGHC and Parent make the covenants and agreements set forth in this Article 5 which covenants and agreements are applicable from this date to the Closing Date.
- 5.2 <u>Consents and Approvals</u>. MGHC and Parent will use their best efforts to cause all conditions to the obligations of MGHC and the Parent hereunder to be satisfied and to obtain all necessary consents and approvals to the performance of its obligations under this Agreement and the transactions contemplated hereby.
- 5.3 <u>Breach of Representations and Warranties</u>. Promptly upon the occurrence of, or promptly upon any of MGHC, the Parent or the MG Merger Entities becoming aware of, the impending or threatened occurrence of any event which would cause or constitute a breach, or would have caused or constituted a breach had such event occurred or been known prior to the date hereof, of any of the representations and warranties of any of MGHC, the Parent or the MG Merger Entities contained in or referred to in this Agreement or in any Schedule hereto, such party shall give detailed written notice thereof to the Nationwide Entities an the Shareholders' Representative.
- 5.4 <u>Consummation of Agreement; Cooperation</u>. Each of MGHC, the Parent and the MG Merger Entities shall use their best efforts, in addition to the performance and fulfillment of all covenants, agreements, conditions and obligations on their parts to be performed and fulfilled under this Agreement, to the end that the Merger Transactions contemplated by this Agreement shall be fully carried by the Closing Date.

## ARTICLE 6. CONDITIONS.

- 6.1 <u>Conditions to the Obligations of MGHC and the Parent</u>. The obligation of MGHC, the Parent and the MG Merger Entities to consummate this Agreement and the transactions contemplated hereby are subject to the fulfillment, prior to or at the Closing, of the following conditions (any one or more of which may be waived in whole or in part by MGHC, the Parent and the MG Merger Entities):
- (a) <u>Representations</u>; <u>Warranties</u>; <u>Covenants</u>. Each of the representations and warranties of the Nationwide Entities and the Shareholders contained in Article 2 shall be true and correct on and as of the Closing Date, with the same effect as though made on and as of the Closing Date; each of the Nationwide Entities and the Shareholders shall, on or before the Closing Date, have performed and satisfied in all material respects all agreements and conditions hereunder which by the terms hereof are to be performed and satisfied by the Nationwide Entities or the Shareholders on or before the Closing Date; each of the Nationwide Entities shall have delivered to the Parent a

32

certificate dated the Closing Date signed on its behalf by its President or its Vice-President to the foregoing effect with respect to the representations, warranties and covenants of the Nationwide Entities, and each Shareholder shall have delivered to the Parent a certificate dated the Closing Date to the foregoing effect with respect to the representations, warranties and covenants of the Shareholder. Notwithstanding the foregoing, the Nationwide Entities may give Parent written notice ("Addendum Notice"), not later than three (3) days prior to the Closing, setting forth any changes to the Schedules or their representations which are necessary to reflect changes to the Nationwide Entities or the Business which occur between the date hereof and the Closing Date, and the Schedules and the representations of the Nationwide Entities and the Shareholders shall be amended by such Addendum Notices as of the Closing Date; provided, however, that no such Addendum Notice shall be effective (i) to reflect any change caused by a breach of any of the covenants of the Nationwide Entities or the Shareholders hereunder, or (ii) to modify Sections or Schedules 2.2, 2.3, 2.4 or 2.19 hereof. The parties recognize and agree that, in the event the Addendum Notice shall modify the representations and warranties in any material way, MGHC, the Parent and the MG Merger Entities shall have the right to terminate this Agreement pursuant to Article 7 hereof.

Each Shareholder shall also have delivered to the Parent a certificate dated the Closing Date certifying that all of such Shareholder's representations and warranties are true and correct in all material respects on and as of the Closing Date, with the same effect as if made on and as of the Closing Date, and that all agreements and conditions hereunder which by the terms hereof are to be performed by such Shareholder on or before the Closing Date have been performed and satisfied in all material respects.

- (b) <u>Approval of Merger Transactions</u>. Each of the Nationwide Entities and all of the Shareholders shall have approved the adoption of this Agreement, the Merger Transactions and any other matters required to be approved by them.
- (c) <u>Debt Financing</u>. The Parent shall have obtained debt financing on terms and conditions satisfactory to the Parent and its board of directors sufficient to pay the Purchase Price and all of the Parent's closing costs, fees and expenses.
- (d) <u>Escrow Agreements</u>. Each of the Shareholders shall have executed and delivered (i) the Escrow Agreement as contemplated in subsection 1.12(b) along with written instructions to deposit their proportionate share of the Escrow Deposit and (ii) the Indemnity Escrow Agreement as contemplated in Section 9.2 along with written instructions to deposit their proportionate share of the Indemnity Escrow Deposit.
- (e) <u>Non-Compete Agreements</u>. Each of the Shareholders shall have executed and delivered a three-year non-compete agreement substantially in form attached hereto as Exhibit J.
- (f) <u>Amendments of Warehouse Lease</u>. The Nationwide Entities shall have amended all leases, contracts, agreements or other arrangements binding on any of the Nationwide Entities relating to the lease, use or occupancy of the three warehouse facilities utilized by the

33

Nationwide Entities as of the date of this Agreement, in each case without penalty, cost or expense to either of the Nationwide Entities (either before or after consummation of the Merger Transactions) and in the form attached hereto as Exhibit K. The three facilities contemplated by this subsection 6.1(f) and the proposed amendments are described on Schedule 6.1(f) attached hereto.

- (g) <u>Distribution of Assets and Related Indebtedness</u>. The Nationwide Entities shall have sold, conveyed, assigned and transferred to the Shareholders or otherwise the Excluded Assets and shall have satisfied, or caused the assignee thereof to assume, and have the Nationwide Entities released from, all related indebtedness, lease obligations, loans or other obligations and burdens as further described on <u>Schedule 6.1(g)</u>, all without penalty, cost or expense to either of the Nationwide Entities or pay to NFR at or promptly after the Closing an amount equal to such indebtedness.
- (h) <u>Opinions of Counsel and Other Documents</u>. On the Closing Date, the Parent shall have received an opinion of counsel for the Nationwide Entities and the Shareholders and addressed to Parent, dated the Closing Date, in the form attached hereto as Exhibit L.
- (i) <u>Certificates</u>. On the Closing Date, the Parent shall have received stock certificates representing all of the issued and outstanding capital stock of each Nationwide Entity.
- (j) <u>Resignations</u>. Each of the Shareholders shall have delivered a resignation, dated the Closing Date, from any and all capacities as a director, officer, partner or agent of each of the Nationwide Entities.
- (k) <u>Current Working Capital</u>. On the Closing Date each of the Shareholders shall have delivered to the Parent a certificate representing that as of the Closing Date the current assets of the Nationwide Entities plus Fifty Thousand Dollars (\$50,000) shall not be less than the current liabilities of the Nationwide Entities as of the Closing Date. "Current assets" and "current liabilities" shall be determined in accordance with GAAP.
- (l) <u>Approvals and Consents</u>. The Nationwide Entities and the Parent shall have received consents from third parties who are the Landlords representing not less than ninety percent (90%) of the leases for the Leased Real Property in the form attached hereto as Exhibit M hereto, such consent not to include any modification to the leases.
- (m) <u>Warranty Re: Affiliate Leases</u>. Each of the Shareholders shall have executed and delivered a Warranty and Maintenance Agreement substantially in the form attached as Exhibit J.
- (n) <u>Release of Liens</u>. Any and all liens, financing statements or encumbrances of any nature burdening any of the assets of the Nationwide Entities shall be released and discharged.

34

- (o) <u>Termination of Affiliated Transactions and Agreements</u>. Each of the agreements and arrangements described on Schedule 6.1(o) shall be terminated, in each case without penalty, cost or expense to either of the Nationwide Entities (either before or after the Merger Transactions) and in such form satisfactory to the Parent and its counsel.
- (p) <u>Tax Letter</u>. The Parent and the Shareholders shall enter into a letter agreement relating to the preparation and filing of all federal and state tax returns for the Nationwide Entities for the 1998 calendar year and for the period during 1999 ending on the Closing Date.
- (q) <u>Accountant's Representation Letter</u>. One or more of the Shareholders, as appropriate, shall have executed and delivered to the Nationwide Accountants all management representation letters and other written assurances reasonably requested by the Nationwide Accountants which are necessary for the Nationwide Accountants to complete the 1998 Balance Sheet and related audited statements of income, stockholders equity and cash flow.
- (r) <u>Miscellaneous Closing Documents</u>. Each of the Nationwide Entities shall have delivered to the Parent a certified copy of its organizational documents, a certificate of existence and good standing, where appropriate, a certified copy of the resolutions approving this Agreement and the transactions contemplated hereby and an incumbency certificate.
- (s) <u>No Actions or Proceedings</u>. No action or proceeding by any court, administrative body or governmental agency shall have been instituted or, to the knowledge of one of the Shareholders or the Nationwide Entities threatened, which would enjoin, restrain or prohibit, or might result in substantial damages to either of the Nationwide Entities in respect of, this Agreement or the complete consummation of the transactions contemplated by this Agreement and no court order shall have been entered in any action or proceeding instituted by any party which enjoins, restrains or prohibits this Agreement or the complete consummation of the transactions as contemplated by this Agreement.
- 6.2 <u>Conditions to the Obligations of the Nationwide Entities and the Shareholders</u>. The obligations of the Nationwide Entities and the Shareholders to consummate this Agreement and the transactions contemplated hereby are subject to the fulfillment, prior to or at the Closing Date, of the following conditions (any one or more of which may be waived in whole or in part by the Representative acting on behalf of the Shareholders):
- (a) <u>Representations</u>; <u>Warranties</u>; <u>Covenants</u>. Each of the representations and warranties of MGHC and the Parent contained in Article 4 shall be true and correct in all material respects on and as of the Closing Date, with the same effect as though made on and as of the Closing Date; MGHC and the Parent shall, on or before the Closing Date, have performed and satisfied in all material respects all agreements and conditions hereunder which by the terms hereof are to be performed and satisfied by MGHC and the Parent on or before the Closing Date; and MGHC and the Parent shall have delivered to the Nationwide Entities a certificate signed on their behalf by their President and dated as of the Closing Date certifying to the foregoing effect.

35

- (b) <u>Merger Filings</u>. The Parent, the MG Merger Entities and the Nationwide Entities shall have made all findings necessary or appropriate to consummate the Merger Transactions, all in accordance with Pennsylvania Law.
- (c) <u>Opinion of Counsel and Other Documents</u>. At the Closing Date, the Shareholders shall have received an opinion of Messrs. Kane, Russell, Coleman & Logan, P.C., counsel for the Parent, dated as of the Closing Date and addressed to the Shareholders, in the form attached hereto as Exhibit N.
- (d) <u>Tax Letter</u>. The Parent and the Shareholders shall enter into a letter agreement relating to the preparation and filing of all federal and state tax returns for the Nationwide Entities for the 1998 calendar year and for the period during 1999 ending on the Closing Date.
- (e) <u>Merger Consideration</u>. The Merger Consideration payable to, or on behalf of, the Shareholders tendering their share certificates has been paid to such Shareholders.
- (f) <u>No Actions or Proceedings</u>. No action or proceeding by any court, administrative body or governmental agency shall have been instituted or threatened which would enjoin, restrain or prohibit, or might result in substantial damages in respect of, this Agreement or the complete consummation of the transactions as contemplated by this Agreement and which would in the reasonable judgment of the Representative acting on behalf of the Shareholders make it inadvisable to consummate such transactions, and no law or regulation shall be in effect and no court order shall have been entered in any action or proceeding instituted by any party which enjoins, restrains or prohibits this Agreement or complete consummation of the transactions as contemplated by this Agreement.
- by the Parent in connection with the consummation of the Closing on the Closing Date and the other transactions contemplated hereby and all certificates, opinions, instruments and other documents required to effect the transactions contemplated hereby reasonably requested by the Nationwide Entities or the Shareholders will be reasonably satisfactory in form and substance to the Nationwide Entities and the Shareholders.
- (h) <u>Executive Employment Agreement</u>. NFR and Andrew Epstein shall have executed and delivered the Executive Employment Agreement in the form attached hereto as Exhibit P.

#### ARTICLE 7. TERMINATION OF AGREEMENT.

- 7.1 <u>Termination</u>. This Agreement may be terminated any time prior to the Closing Date as follows:
  - (a) With the mutual consent of MGHC, the Parent and the Shareholders.

36

- By delivery of written notice of termination to the Representative by MGHC and the Parent, (i) if there has been a misrepresentation or breach of warranty on the part of any of the Nationwide Entities or a Shareholder in the representations and warranties contained herein, (ii) if there has been a breach of any agreement or covenant on the part of any of the Nationwide Entities or a Shareholder and the same has not been cured within 5 days after notice thereof or (iii) if the conditions precedent set forth in Section 6.1 have not been satisfied. In the event of any termination pursuant to this subsection 7.1(b), written notice setting forth the reasons therefor shall forthwith be given by MGHC and the Parent to the Nationwide Entities and the Shareholders. If this Agreement is terminated by MGHC and the Parent pursuant to this Section 7.1(b), neither MGHC, the Parent and the MG Merger Entities on the one hand nor the Nationwide Entities and the Shareholders on the other hand shall have any liability to each other except for those provisions of this Agreement that by their terms survive such termination and except in the event that the misrepresentation or breach giving rise to such termination arises from fraud on the part of any of the Nationwide Entities or a Shareholder, breach by any of the Nationwide Entities or a Shareholder of its or his obligations under Section 3.5 or intentional breach by the Nationwide Entities or any Shareholder of any other provision of this Agreement for the purpose of enabling the Nationwide Entities or the Shareholders, or any of them, to evade or avoid their obligations hereunder.
- By delivery of written notice of termination to MGHC and the Parent from the Representative acting on behalf of the Nationwide Entities and the Shareholders, (i) if there has been a misrepresentation or breach of warranty on the part of MGHC, the Parent or the MG Merger Entities in the representations and warranties contained herein, (ii) if there has been a breach of any agreement or covenant on the part of MGHC or the Parent and the same has not been cured within 5 days after notice thereof or (iii) if the conditions precedent set forth in Section 6.2 have not been satisfied. In the event of any termination pursuant to this subsection 7.1(c), written notice setting forth the reasons therefor shall forthwith be given by the Nationwide Entities to MGHC and the Parent. If this Agreement is terminated pursuant to this subsection 7.1(c), neither MGHC, the Parent and the MG Merger Entities on the one hand nor the Nationwide Entities and the Shareholders on the other hand shall have any liability to each other except for those provisions of this Agreement that, by their terms, survive such termination and except in the event that the misrepresentation or breach of warranty or covenant giving rise to such termination arises from fraud on the part of MGHC, the Parent or the MG Merger Entities or intentional breach by MGHC, the Parent or the MG Merger Entities of any provision of this Agreement for the purpose of evading or avoiding its obligations hereunder.
- 7.2 <u>Effect of Termination</u>. All obligations of the parties hereunder shall cease upon any termination pursuant to Section 7.1, provided, however, that (i) the provisions of this Article 7 and Section 8.2 hereof shall survive any termination of this Agreement in any event, and (ii) any party may proceed as further set forth in Section 7.3 below.
- 7.3 <u>Right to Proceed</u>. Anything in this Agreement to the contrary notwithstanding, if any of the conditions specified in Section 6.1 hereof has not been satisfied, MGHC and the Parent shall have the right to proceed with the transactions contemplated hereby and if any of the conditions

specified in Section 6.2 hereof has not been satisfied, the Nationwide Entities and the Shareholders shall have the right to proceed with the transactions contemplated hereby. In the event the Parent, the Nationwide Entities and/or the Shareholders proceed with the transactions contemplated hereby, all conditions which have not been satisfied shall be deemed waived.

#### ARTICLE 8. CERTAIN RIGHTS AND OBLIGATIONS SUBSEQUENT TO CLOSING.

- 8.1 <u>Survival of Warranties</u>. All representations, warranties, agreements, covenants and obligations herein or in any Schedule or certificate delivered by any party incident to the transactions contemplated hereby are material and may be relied upon by the party receiving the same and shall survive the Closing for a period of twelve (12) months (except for the representations and warranties set forth in Sections 2.4 and 2.8 which shall survive the Closing for a period of forty-eight (48) months) regardless of any investigation by or knowledge of such party and shall not merge into the performance of any obligation by any party hereto, subject to the provisions of Article 9 hereof.
- 8.2 <u>Fees and Expenses</u>. Each party shall pay its own expenses and costs incidental to the preparation of this Agreement and to the consummation of the transactions contemplated hereby; provided that the Nationwide Entities shall pay all fees associated with their transactions applicable to themselves and the Shareholders which are accrued prior to the Closing Date.

#### ARTICLE 9. INDEMNIFICATION.

9.1 Indemnification by the Shareholders. Subject to the minimum amount and other limitations and provisions set forth in Section 9.2 if and to the extent applicable as provided therein, the Shareholders, jointly and severally (except as to the representations in subsections 2.4(b) and 2.5(b) and (c) which shall be several only), agree subsequent to the Closing Date to indemnify and hold harmless MGHC, the Parent, the Nationwide Entities and their respective affiliates, officers, directors, employees and agents (individually, a "Parent Indemnified Party" and collectively, the "Parent Indemnified Parties") from and against and in respect of all losses, liabilities, obligations, damages, deficiencies, actions, suits, proceedings, demands, assessments, orders, judgments, fines, penalties, costs and expenses (including the reasonable fees, disbursements and expenses of attorneys, accountants and consultants) of any kind or nature whatsoever (whether or not arising out of third-party claims and including all amounts paid in defense or settlement of the foregoing) sustained, suffered or incurred by or made against any Parent Indemnified Party (a "Loss" or "Losses") arising out of (a) any breach of any representation or warranty made by any of the Nationwide Entities or any Shareholder in this Agreement or in any Schedule, exhibit, certificate, agreement or other instrument delivered under or in connection with this Agreement; (b) any breach of any covenant or agreement made by any of the Nationwide Entities or any Shareholder in this Agreement or in any Schedule or exhibit hereto, or in any certificate, agreement or other instrument delivered at the Closing under or in connection with this Agreement; and/or (c) the sale by NFR of its rental and leasing business pursuant to that certain Asset Purchase Agreement dated September 30, 1998 by and between NFR and American Furniture Rentals, Inc.

38

STARKP/AGR/134740\_7(16480/03)

TRADEMARK
REEL: 002890 FRAME: 0916

With respect to each Loss indemnifiable by the Shareholders pursuant to this Section 9.1 (each a "Parent Indemnifiable Loss"), in the event any Shareholder is liable for a Parent Indemnifiable Loss referred to in this Section 9.1, such Shareholder agrees not to seek indemnity against the Nationwide Entities, or any of them, as a result of any rights to indemnification provided under the articles of incorporation, bylaws, or any other document, agreement or instrument executed by, or binding on, any such Nationwide Entity, or the statutes of any state governing the Nationwide Entities except to the extent of insurance coverage therefor.

## 9.2 <u>Limitations on Indemnification by the Shareholders.</u>

- Parent Indemnified Parties except to the extent the cumulative amount of Parent Indemnifiable Losses exceeds Three Hundred Thousand Dollars (\$300,000) (the "Minimum Threshold") provided, however, such Minimum Threshold shall not be applicable to any Parent Indemnifiable Losses which occur as a result of fraud, any intentional misrepresentation of a material fact made by any Shareholder or any breach of the representations contained in the certificate to be delivered at the Closing pursuant to subsection 6.1(k) hereof. The Shareholders agree to escrow at the Closing, in proportion to the Merger Consideration payable to them as set forth in Exhibit B, an aggregate of One Million Nine Hundred Sixty Thousand Dollars (\$1,960,000.00) of the Purchase Price (the "Indemnity Escrow Deposit") pursuant to an Indemnity Escrow Agreement substantially in the form attached hereto as Exhibit O, such escrow to insure the availability of initial funds to support the indemnity of the Shareholders contemplated herein.
- (b) <u>Time Limits for Claims</u>. Other than indemnification with respect to the Losses as a result of any misrepresentation or breach of warranty set forth in Sections 2.4 and 2.8 which shall expire on the date forty-eight (48) months following the Closing Date, indemnification with respect to all Parent Indemnifiable Losses shall expire on the date twelve (12) months following the Closing Date; provided further, however, that in each case, if prior to the applicable date of expiration a specific claim for indemnification for a Parent Indemnifiable Loss, a Parent Indemnified Party shall have given notice of such claim to the Shareholders' Representative, then the right to indemnification with respect thereto shall remain in effect until such matter shall have been finally determined and disposed of, and any indemnification due in respect thereof shall have been paid, according to the date on which notice of the applicable claim is given.
- (c) <u>Sole Remedy</u>. Indemnification pursuant to Section 9.1 is the sole remedy of Parent Indemnified Parties against the Shareholders for matters arising out of the transactions contemplated by this Agreement after the Closing. Notwithstanding anything herein to the contrary, the limitations set forth in this Section 9.2 shall in no way limit any rights the Parent may have in law or equity, consistent with Section 7.2, in the event the Closing does not occur.

39

## 9.3 <u>Indemnification by MGHC and the Parent.</u>

- Shareholders and their respective officers, directors, employees and agents (individually, a "Seller Indemnified Party" and collectively, "Seller Indemnified Parties") from and against and in respect of all losses, liabilities, obligations, damages, deficiencies, actions, suits, proceedings, demands, assessments, orders, judgments, fines, penalties, costs and expenses (including the reasonable fees, disbursements and expenses of attorneys, accountants and consultants) of any kind or nature whatsoever (whether or not arising out of third party claims and including all amounts paid in investigation, defense or settlement of the foregoing) sustained, suffered or incurred by or made against any Seller Indemnified Party arising out of (a) any breach of any representation or warranty made by the Parent in this Agreement or in any Schedule, exhibit, certificate, agreement or other instrument delivered under or in connection with this Agreement, and (b) any breach of any covenant or agreement made by Parent in this Agreement or in any Schedule, exhibit, certificate, agreement or other instrument delivered under or in connection with this Agreement (such claims under clauses (a) and (b) being hereinafter collectively referred to as "Seller Indemnifiable Losses").
- (b) Indemnification with respect to all Seller Indemnifiable Losses contemplated by subsections 9.3(a) shall expire on the date twelve (12) months following the Closing Date; provided, however, that in each case, if prior to the applicable date of expiration a specific claim shall have become known which may constitute or give rise to any Seller Indemnifiable Loss as to which indemnity may be payable and a Seller Indemnifiable Party shall have given notice of such claim to the Parent, then the right to indemnification with respect thereto shall remain in effect until such matter shall have been finally determined and disposed of, and any indemnification due in respect thereof shall have been paid, according to the date on which notice of the applicable claim is given.

## 9.4 Notice; Defense of Claims.

(a) Promptly after receipt by an indemnified party of notice of any claim, liability or expense to which the indemnification obligations hereunder would apply, the indemnified party shall give notice thereof in writing to the indemnifying party (the Parent with respect to claims by Seller Indemnified Parties and the Representative with respect to claims by Parent Indemnified Parties), but the omission to so notify the indemnifying party promptly will not relieve the indemnifying party from any liability except to the extent that the indemnifying party shall have been prejudiced as a result of the failure or delay in giving such notice. Such notice shall state the information then available regarding the amount and nature of such claim, liability or expense and shall specify the provision or provisions of this Agreement under which the liability or obligation is asserted. If within twenty days after receiving such notice the indemnifying party gives written notice to the indemnified party stating that it disputes and intends to defend against such claim, liability or expense at its own cost and expense, then counsel for the defense shall be selected by the indemnifying party and the indemnifying party shall not be required to make any payment with respect to such claim, liability or expense as long as the indemnifying party is conducting a good

40

faith and diligent defense at its own expense; provided, however, that the assumption of defense of any such matters by the indemnifying party shall relate solely to the claim, liability or expense that is subject or potentially subject to indemnification. Each indemnified party shall cooperate in all reasonable respects, at the indemnifying party's request and cost, risk and expense, with the indemnifying party and its attorneys in the investigation, trial and defense of such claim, liability or expense and any resulting suit, proceeding or enforcement action and any appeal therefrom. The indemnifying party shall have the right, with the consent of the indemnified party, which consent shall not be unreasonably withheld, to settle all indemnifiable matters related to claims by third parties which are susceptible to being settled, provided that no such consent shall be required if as part of such settlement the indemnified party is released from all liability relating to the subject matter of such settlement. The indemnifying party shall keep the indemnified party apprised of the status of the claim, liability or expense and any resulting suit, proceeding or enforcement action, shall furnish the indemnified party with all documents and information that the indemnified party shall reasonably request and shall consult with the indemnified party prior to acting on major matters, including settlement discussions. Notwithstanding anything herein stated, the indemnified party shall at all times have the right to fully participate in such defense at its own expense directly or through counsel; provided, however, if the named parties to the action or proceeding include both the indemnifying party and the indemnified party and representation of both parties by the same counsel would be inappropriate under applicable standards of professional conduct, the expense of separate counsel for the indemnified party shall be paid by the indemnifying party. If no such notice of intent to dispute and defend is given by the indemnifying party, or if such defense is not being or ceases to be conducted, the indemnified party shall, at the expense of the indemnifying party, undertake the defense of (with counsel selected by the indemnified party), and shall have the right to compromise or settle (exercising reasonable business judgment), such claim, liability or expense. If such claim, liability or expense is one that by its nature cannot be defended solely by the indemnifying party, then the indemnified party shall make available all information and assistance that the indemnifying party may reasonably request and shall cooperate with the indemnifying party in such defense.

- 9.5 <u>Satisfaction of Shareholders' Indemnification Obligations</u>. Any indemnity payable pursuant to this Section 9 shall be paid within the later of (a) thirty (30) days after the indemnified party's request therefor or (b) ten (10) days prior to the date on which the Loss upon which the indemnity is based is required to be satisfied by the indemnified party.
- 9.6 Arbitration of Claims. In the event any of the Parent Indemnified Parties or the Seller Indemnified Parties shall seek indemnity hereunder against any other party to this Agreement for a Parent Indemnified Loss or a Seller Indemnified Loss, respectively, the parties agree to resolve any claim for such loss within the thirty (30) day period following the delivery of a notice of claim to the indemnifying party. In this event the parties cannot resolve any claim for such loss as set forth in such notice of claim during such thirty (30) day period, such claim shall be submitted for arbitration by the AAA in Philadelphia, Pennsylvania, the expense of which shall be shared equally by the Parent and the Shareholders, provided that each party shall bear its own expenses. The parties shall request the AAA to appoint three arbitrators for resolution of such claim (unless the amount subject

41

to the claim is \$50,000 or less in which case it shall be one arbitrator) whose decision shall be final. Any amounts due pursuant to the arbitrators' decision shall be paid within ten days of their decision.

9.7 Special Provisions Relating to Subsection 6.1(k). Notwithstanding anything to the contrary contained herein, the Parent shall notify in writing the Representative, within sixty (60) days of the Closing Date, of any breach claimed by the Parent of the Shareholders' representations contained in the certificate to be delivered pursuant to Section 6.1(k) (the "Section 6.1(k) Notice"). The Parent and the Representative shall attempt to resolve any dispute regarding the Section 6.1(k) Notice within thirty (30) days following delivery of such Section 6.1(k) Notice. In the event there shall continue to remain any dispute following such thirty (30) days, such dispute shall be submitted for arbitration by the AAA in Philadelphia, Pennsylvania, the expense of which shall be shared equally by the Parent and the Shareholders, provided that each party shall bear its own expenses. The parties shall request the AAA to appoint three arbitrators for such dispute (unless the amount in dispute is \$50,000 or less in which case it shall be one arbitrator) whose decision shall be final. Any amounts due pursuant to the arbitrators' decision shall be paid within ten days of their decision.

#### ARTICLE 10. MISCELLANEOUS.

- 10.1 <u>Law Governing</u>. This Agreement shall be construed under and governed by the laws of the Commonwealth of Pennsylvania without regard to its conflict of laws provisions.
- 10.2 <u>Notices</u>. Any notice, request, demand other communication required or permitted hereunder shall be in writing and shall be deemed to have been given if delivered personally, upon receipt, or if sent by registered or certified mail upon the sooner of the expiration of three days after deposit in United States post office facilities properly addressed with postage prepaid or acknowledgment of receipt or if sent by any national overnight delivery services, one business day after delivery to such service. All notices and payments to a party will be sent to the addresses set forth below or on or to such other address or person as such party may designate by notice to each other party hereunder:

#### TO MGHC OR THE PARENT:

Phil Lang 14665 Midway Road, Suite 100 Dallas, Texas 75244 Facsimile No: 972/392-7308

42

STARKP/AGR/134740\_7(16480/03)

TRADEMARK
REEL: 002890 FRAME: 0920

with a copy to:

Patrick V. Stark
Kane, Russell, Coleman & Logan, P.C.
3700 Thanksgiving Tower
1601 Elm Street
Dallas, Texas 75201
Facsimile No: 214/777-4299

#### TO THE NATIONWIDE ENTITIES:

Nationwide Furniture Rentals & Sales, Inc. 3800 Frankford Ave.
Philadelphia, Pennsylvania
Facsimile No: 610/743-9179

### with a copy to:

A. John May III Pepper Hamilton LLP 1235 Westlakes Drive Berwyn, Pennsylvania 19312 Facsimile No: 610/640-7835

#### TO SHAREHOLDERS' REPRESENTATIVE:

Kenneth F. Mazda 2050 Grantham Road Berwyn, Pennsylvania 19312 Facsimile No. 610/648-0859

#### TO SHAREHOLDERS:

At their address and/or facsimile number set forth on the signature page hereto and to the Shareholders' Representative

Any notice given hereunder may be given on behalf of any party by its counsel or other authorized representative.

10.3 <u>Entire Agreement</u>. This Agreement, including the Schedules and Exhibits referred to herein and the other writings specifically identified herein or contemplated hereby or delivered in connection with the transactions contemplated hereby, is complete, reflects the entire agreement of the parties with respect to its subject matter, and supersedes all previous written or oral

43

STARKP/AGR/134740\_7(16480/03)

TRADEMARK REEL: 002890 FRAME: 0921 negotiations, commitments and writings including the letter of intent dated July 8, 1998. Notwithstanding the foregoing, the parties agreed that the Confidentiality and Non-Compete Agreement dated July 15, 1998 shall not terminate but shall remain in full force and effect as if this Agreement had not been executed.

- 10.4 <u>Assignability</u>. This Agreement shall be assignable by the Parent prior to the Closing only to a wholly-owned subsidiary of the Parent although no such assignment shall relieve the Parent of any liabilities or obligations under this Agreement. This Agreement (including without limitation the provisions of Article 9) shall be binding upon and enforceable by, and shall inure to the benefit of, the parties hereto and their respective successors, heirs, executors, administrators and permitted assigns, and no others. This Agreement may not be assigned by the Nationwide Entities or the Shareholders, except the heirs, executors and administrators of a deceased Shareholder.
- 10.5 <u>Captions and Gender</u>. The captions in this Agreement are for convenience only and shall not affect the construction or interpretation of any term or provision hereof. The use in this Agreement of the masculine pronoun in reference to a party hereto shall be deemed to include the feminine or neuter pronoun, as the context may require.
  - 10.6 Certain Definitions. For purposes of this Agreement, the term:
- (a) "affiliate" of a person shall mean a person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, the first mentioned person;
- (b) "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly or as trustee or executor, of the power to direct or cause the direction of the management policies of a person, whether through the ownership of stock, as trustee or executor, by contract or credit arrangement or otherwise;
- (c) "person" means an individual, corporation, partnership, association, trust or any unincorporated organization; and
- (d) "subsidiary" of a person means any corporation more than fifty percent of whose outstanding voting securities, or any partnership, joint venture or other entity more than fifty percent of whose total equity interest, is directly or indirectly owned by such person.
- 10.7 <u>Execution in Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which shall constitute one and the same document.
- 10.8 <u>Amendments: Waivers</u>. This Agreement may not be amended or modified, nor may compliance with any condition or covenant set forth herein be waived, except by a writing duly and validly executed by the Parent, the Nationwide Entities and the Representative, acting on behalf of

44

the Shareholders, or, in the case of a waiver, the party waiving compliance. No delay on the part of any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any waiver on the part of any party of any such right, power or privilege, or any single or partial exercise of any such right, power or privilege, preclude any further exercise thereof or the exercise of any other such right, power or privilege.

- 10.9 <u>Consent to Jurisdiction</u>. Solely for the purpose of allowing a party to enforce its indemnification and other rights hereunder, each of the parties, on his, her or its behalf and on behalf of its successors and assigns, hereby consents to personal jurisdiction, service of process and venue in the federal or state courts of Pennsylvania or in the court in which any claim for which indemnification may be sought hereunder is brought against an indemnified party.
- 10.10 <u>Deposit Agreement</u>. Upon execution of this Agreement, the Parent and the Representative and the Nationwide Entities shall enter into and deliver that certain Deposit Agreement in the form attached hereto as Exhibit 10.10 pursuant to which the Parent will deposit in escrow the sum of Two Hundred Thousand Dollars (\$200,000), sum deposit to be held pursuant to the terms of such Deposit Agreement.

#### ARTICLE 11. CERTAIN TAX MATTERS.

- Entities, which is a subchapter S corporation, and the Shareholders will join with Parent in making an election under Code § 338(h)(10) of the Code (and any corresponding election under state, local, and foreign tax law) with respect to the Merger Transactions hereunder (a "Section 338(h)(10) Election"). Each Shareholder will include any and all income, gain, loss, deduction, or other tax item resulting from the Section 338(h)(10) Election with respect to the Merger Transaction on his Tax return to the extent permitted by applicable law and neither Parent, the MG Entities nor the Nationwide Entities shall be liable or responsible for any such Taxes. Each Shareholder shall also pay any Tax imposed on the Nationwide Entities attributable to the making of the Section 338(h)(10) Election, including, but not limited to, (i) any Tax imposed under Code § 1374, (ii) any tax imposed under Reg. § 1.338(h)(10)-1(e)(5), or (iii) any state, local or foreign Tax imposed on the Nationwide Entities' gain, and each Shareholder shall indemnify the Parent, the MG Entities and the Nationwide Entities against any adverse consequences arising out of any failure to pay any such Taxes.
- 11.2 <u>Allocation of Purchase Price</u>. The Parent, the MG Entities and the Nationwide Entities agree that the Merger Consideration will be allocated to the assets of the Nationwide Entities for all tax purposes as shown on the allocation schedule attached hereto as Exhibit G. The Parent, the MG Entities and the Nationwide Entities will file all Tax returns (including amended returns and claims for refund) and information reports in a manner consistent with such allocation.
- 11.3 <u>S Corporation Status</u>. Prior to the Closing, the Nationwide Entities and the Shareholders will not revoke the Nationwide Entities' election to be taxed as an S corporation

45

within the meaning of Code §§ 1361 and 1362 and the Nationwide Entities and the Shareholders will not take any action other than the Merger Transactions hereunder that would result in the termination of the Nationwide Entities' status as a validly electing S corporation within the meaning of Code §§ 1361 and 1362.

Parent shall prepare or cause to be prepared and file or cause to be filed all Tax returns for the Nationwide Entities for all periods ending on or prior to the Closing Date which are filed after the Closing Date. The tax returns for the 1998 calendar year and the period between the 1998 Balance Sheet and the Closing Date shall be prepared consistent with the audited financial statements for such periods. The Parent shall be responsible for the initial preparation of such returns and shall permit the Representative to review and comment on each such Tax return described in the preceding sentence prior to filing. To the extent permitted by applicable law, the Shareholders shall include any and all income, gain, loss, deduction or deduction or other tax items for such periods on their Tax returns in a manner consistent with the Schedule K-1s prepared for the Shareholders for such periods.

#### 11.5 Cooperation on Tax Matters.

- (a) The Parent, the MG Entities and Nationwide Entities shall cooperate fully, as and to the extent reasonably requested by any other party, in connection with the filing of Tax returns pursuant to this Section 11 and any audit, litigation or other proceeding with respect to such Taxes. Such cooperation shall include the retention and (upon the other party's request) the provisions of records and information which are reasonably relevant of any such audit, litigation or other proceeding and making employees available to provide additional information and explanation of any material provided hereunder. The Parent agrees (A) to retain all books and records with respect to Tax matters pertinent to the Nationwide Entities relating to any taxable period beginning before the Closing Date until the expiration of the statute of limitations (and any extensions thereof) of the respective taxable periods, and (B) to give Shareholders copies of such books and records.
- (b) The Parent and the Shareholders further agree, upon request, to use their best efforts to obtain any certificate or other document from any governmental authority or any other person as may be necessary to mitigate, reduce or eliminate any Tax that could be imposed (including, but not limited to, with respect to the transactions contemplated hereby).
- (c) To the extent that the Parent and the Shareholders are unable to agree as to the appropriate tax treatment of any item relating to the business of the Nationwide Entities prior to the Closing Date, the Parent and the Shareholders agree to submit any dispute relating thereto to arbitration in Philadelphia, Pennsylvania.
- 11.6 <u>Post-Closing Audits</u>. In the event that after the Closing there shall be commenced any audit of either of the Nationwide Entities by the Internal Revenue Service or any third party

46

regulatory authority, the Parent agrees that (i) the Shareholders and their tax advisors shall be given reasonable access during normal business hours at the offices of the Nationwide Entities to the books and records of the Nationwide Entities applicable to such audit, and (ii) with respect to audits relating to periods prior to the Closing Date, the Shareholders shall be entitled to control, direct and resolve any such audit and the Parent agrees to cooperate with and assist the Shareholders in such audit proceedings including naming the Representative as attorney-in-fact to deal with the IRS provided that the Shareholders shall pay and reimburse any tax owed by the Nationwide Entities (and not passed on to the Shareholders) as a result of such audit of a period prior to the Closing Date.

## ARTICLE 12. POST-CLOSING COVENANTS.

Assistance in Accounting Matters. From and after the Closing Date, the Representative, Stephen Mazda and any other Shareholder who executes a management representation letter in connection with the 1998 Balance Sheet agree to provide, upon request of the Parent, reasonable assistance to the Parent, the Nationwide Accountants and Deloitte & Touche, LLP (the Parent's accountants) in preparing and finalizing (i) the 1998 Balance Sheet and related audited statements of income, stockholders' equity and cash flow, and (ii) an audited balance sheet as of the Closing Date (or other date close to the Closing Date which is selected by the Parent) and related audited statements of income, stockholders' equity and cash flow for the period from the 1998 Balance Sheet date to the date of the balance sheet contemplated in this subsection (ii). Such reasonable assistance shall include assistance with information requests and execution and delivery of appropriate management representation letters and other written assurances requested by the accountants conducting such audit; provided however, such reasonable assistance regarding information requests shall not require any such party to travel or to perform any duties other than answer questions of the Parent as to financial matters relating to the Nationwide Entities prior to the Closing Date to which he has knowledge.

47

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first set forth above.

#### MATTRESS GIANT HOLDING CORPORATION

By: Its: **MATTRESS GIANT ACQUISITION CORPORATION** By: Its: M G MERGER SUB NO. 1, INC. By: Its: M G MERGER SUB NO. 2, INC. By: Its: NATIONWIDE FURNITURE RENTALS & SALES, INC.

48

By: Its:

# By: Its: **SHAREHOLDERS:** Kenneth F. Mazda Steven Mazda Ron D'Argenio

NATIONWIDE MATTRESS GIANT, INC.

## SCHEDULE 2.13

# [Proprietary Rights]

# (1) Proprietary Rights

- -Registered Trademark "Nationwide Discount Sleep Centers"
- -Licenses of various computer software packages for accounting and other uses.

#### (2) <u>Title</u>

-NFR has licensed its rights, if any, to the name "Nationwide Furniture Rentals" to American Furniture Rentals, Inc.

TRADEMARK REEL: 002890 FRAME: 0928

RECORDED: 12/24/2003